

Contents lists available at ScienceDirect

Journal of Systems Architecture



journal homepage: www.elsevier.com/locate/sysarc

A comprehensive systematic review of integration of time sensitive networking and 5G communication

Zenepe Satka^{*}, Mohammad Ashjaei, Hossein Fotouhi, Masoud Daneshtalab, Mikael Sjödin, Saad Mubeen

Mälardalen University, Sweden

A – Ka Ti TS 50 UI In

RTICLE INFO	A B S T R A C T
ywords: me-sensitive networking N N RLC dustry 4.0	Many industrial real-time applications in various domains, e.g., automotive, industrial automation, industrial IoT, and industry 4.0, require ultra-low end-to-end network latency, often in the order of 10 milliseconds or less. The IEEE 802.1 time-sensitive networking (TSN) is a set of standards that supports the required low-latency wired communication with ultra-low jitter. The flexibility of such a wired connection can be increased if it is integrated with a mobile wireless network. The fifth generation of cellular networks (5G) is capable of supporting the required levels of network latency with the Ultra-Reliable Low Latency Communication (URLLC) service. To fully utilize the potential of these two technologies (TSN and 5G) in industrial applications, seamless integration of the TSN wired-based network with the 5G wireless-based network is needed. In this article, we provide a comprehensive and well-structured snapshot of the existing research on TSN-5G integration. In this regard, we present the planning, execution, and analysis results of the systematic review. We also identify the trends, technical characteristics, and potential gaps in the state of the art, thus highlighting future research directions in the integration of TSN and 5G technologies, introducing approaches with an accuracy starting from the levels of hundred nanoseconds to one microsecond. Majority of primary studies aim at optimizing communication latency in their approach, which is a key quality attribute in automotive and industrial automation applications today.

1. Introduction

There is an urgent need to support ultra-reliable, high-bandwidth, low-latency and predictable communication in many contemporary and future industrial applications [1,2]. Examples of these applications include autonomous driving, autonomous construction sites and mines, collaborating robots, augmented and virtual reality, to mention a few [3–5]. The end-to-end communication in these applications is achieved by combining wired networks (for onboard communication) and wireless networks (for remote communication). The end-toend (E2E) latency in these applications often ranges from milliseconds for wired onboard communication to microseconds for wireless communication [6,7].

Time-sensitive networking (TSN) is a set of standards for Ethernetbased communication, where the main focus is on providing lowlatency and low-jitter for time-sensitive traffic, and even to provide deterministic message transmission over switched Ethernet [6,8–11]. Since it is a wired network, it limits the connection only to the areas where a wired connection is feasible while missing the flexibility of a mobile connection.

The fifth generation of wireless telecommunications (5G) [12,13] can provide the flexibility and scalability required of the mentioned applications. 5G supports real-time applications by using a service called Ultra-Reliable Low Latency Communication (URLLC) which is part of the 3rd Generation Partnership Project (3GPP) Releases [14–17]. URLLC is a promising candidate for real-time wireless communication that can support latency down to 1 ms and reliability up to 99.999% [18–20].

A converged wired and wireless network integrating TSN and 5G is needed to achieve real-time requirements, determinism as well as mobility and scalability of communication in contemporary and future industrial applications. Moreover, the integration of TSN with 5G offers improved Quality of Service (QoS) for time-sensitive applications, such as autonomous driving, industrial automation, and virtual reality. By integrating TSN with 5G, industries will take advantage of the benefits of both technologies to enable new use cases and applications. To

https://doi.org/10.1016/j.sysarc.2023.102852

Received 27 September 2022; Received in revised form 27 January 2023; Accepted 20 February 2023 Available online 26 February 2023

1383-7621/© 2023 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

^{*} Corresponding author. E-mail address: zenepe.satka@mdu.se (Z. Satka).

achieve the E2E QoS requirements of a TSN-5G network a significant effort is required due to the large dissimilarity of the considered systems.

Several core challenges are encountered when integrating the TSN and 5G technologies. Some notable challenges include understanding of the different architectural trade-offs in a joint TSN-5G architecture, time synchronization between the two technologies, simulation, and implementation of a TSN-5G network in a real-world environment, among others. Various initiatives aimed at addressing these challenges. For example, the overview of relevant architecture aspects and the relevant features and processes are described in 3GPP TS 23.501, TS 23.502, TS 23.503, and 5G-ACIA [18]. The interaction between the two technologies is part of 3GPP specifications and there are many research works in this regard. In this paper, we construct a structured map of the existing literature on the integration of TSN and 5G communication technologies. We show that there exist several gaps in the state of the art that require the immediate attention of the community to achieve a seamless integration of TSN and 5G for industrial applications.

1.1. Paper contributions

This Systematic Literature Review (SLR) identifies the research studies conducted in the area of TSN and 5G integration. The main goal of this SLR is to provide a detailed investigation of state of the art on TSN-5G integration and provide a fully-concentrated and wellorganized classification scheme introduced in Section 3. This will help the researchers and practitioners in identifying and understanding the existing solutions and their applicability in industrial environments. Another contribution of this SLR is the identification of gaps in the current research and highlighting the opportunities for further research in the area of TSN-5G integration.

In this article, from an initial set of 189 studies, we identified 82 primary studies.¹ We analyzed these studies in detail, following a structured data extraction, analysis, and synthesis process. A summary of the resulting highlights of our study is as follows:

- The efforts in this research area started in 2018, after the initial delivery of 3GPP Release 15 in late 2017.
- 73% of the primary studies address time synchronization between the two technologies, which still represents a significant challenge in the integration of these two technologies.
- In the context of time synchronization between the two technologies, the transparent clock approach is mostly preferred over the boundary clock approach.
- 74% of the primary studies follow an integration architecture, which conforms to the 3GPP releases.
- Majority of primary studies aim at optimizing communication latency in their approach, which is a key quality attribute in automotive and industrial automation applications today.
- Most of the primary studies remain generic without focusing on a specific domain.
- The technical contributions provided by the majority of the existing studies are focused on the integration architecture of TSN and 5G. In this regard, the majority of these studies provide solution proposals or validation research, while missing other types of research such as evaluation research, or experience papers. None of the primary studies have provided a tool as a contribution. This indicates that there is an urgent need for provisioning of tools that incorporate the existing techniques or new techniques for TSN-5G integration.

1.2. Paper outline

The rest of the paper is organized as follows. Section 2 describes the research method in detail. Section 3 provides a detailed overview of our data extraction form. Section 4 shows the vertical results of our study, while Section 5 illustrates the horizontal results. Section 6 presents Fleiss Kappa statistical analysis that we used to mitigate threats to validity. In Section 7, we discuss the potential threats to the validity of our study and how we mitigate them. Section 8 presents similar works to our study. Section 9 concludes the paper and presents the future work.

2. Study selection process

This study is conducted and fulfilled based on the well-known guidelines presented in [21,22]. The process is divided into 3 main phases: (i) planning, (ii) conducting, and (iii) documenting as shown in Fig. 1.

Planning. The main objective in the planning phase is to identify the research questions (RQs) and the need for a review of related works and approaches that are performed within the scope of TSN-5G integration. From this phase, a detailed protocol was defined following the specified steps to conduct the study systematically.

Conducting. This phase starts with the search and selection step, where the automatic string search is performed in the four largest databases hosting the research in the domains of computer science, computer engineering, software engineering, and systems engineering, among others. Based on the authors' knowledge of the targeted research domain, three primary studies [23–25] were chosen from the search pool to be considered in the extraction form of the review process. A set of parameters were identified for the classification scheme that was used for the data extraction form of our study. We specified those parameters by systematically applying the standard keywording process [26]. After fulfilling the data extraction form for each of the research studies, we analyzed and synthesized the extracted data to address the research questions (see Section 2.1) posed in this SLR.

Documenting. In this phase, we carried out a detailed analysis of the extracted data. Furthermore, we identified possible threats to the validity of our study. A comprehensive analysis was performed for threat validation and verification. This article was written to document and illustrate the performed study in detail.

2.1. Research goal and questions

The goal of this SLR is:

"to classify the technical characteristics as well as to investigate research trends, identify gaps in the state of the art, and highlight open challenges and future directions in the research on integration of TSN and 5G technologies for end-to-end communication in the industrial applications."

To achieve this goal, the following research questions are identified based on the authors' knowledge of the research area:

- **RQ1:** What are the technical characteristics of TSN and 5G integration? **Objective:** to identify and classify existing approaches and techniques for TSN and 5G integration in terms of their technical characteristics.
 - RQ1.1: How is resource management conducted in TSN-5G integrated architectures? Objective: to classify studies in terms of resource manage-
 - ment aspects like configuration model and scheduling. **RQ1.2:** How is the traffic flow managed between TSN and 5G networks?

¹ Primary studies refer to the state of the art publications that are relevant to the goal of this SLR study.

Objective: to classify studies related to the different protocols and models they use to manage the traffic flow.



Fig. 1. An overview of the SLR process.

- **RQ1.3:** Which time synchronization technique is used to achieve a deterministic network environment? **Objective:** to classify the studies in terms of their time synchronization approaches.
- RQ1.4: Which integration architectures are proposed by the research community and what kind of properties do they present?Objective: to identify different integration approaches and classify the studies regarding the properties they address.
- **RQ2:** What are the publication trends of research on TSN and 5G integration?

Objective: to classify a set of relevant studies in order to assess trends and venues over time.

RQ3: What are the limitations of TSN-5G integration?

Objective: to identify current research gaps and limitations with respect to the state of the art research on TSN-5G integration.

The answer to RQ1 will require a detailed investigation of the technical part of this integration, specifically in answering the following questions: (i) what are the different aspects of this integration? (ii) how two different networks can communicate with each other to have a fully converged network? And (iii) what are the core challenges identified in the existing literature. The answer to RQ2 is expected to provide a detailed overview of the current publication trends, research types, and venues. Finally, the answer to RQ3 will assist the research community in understanding whether there is room for improvement in the existing approaches and further research opportunities in this research area.

2.2. Search and selection strategy

After defining the research goal and research questions of our study, we gather the relevant studies available in the research area as presented in Fig. 2.

Before proceeding with the search pool, we first manually select a set of pilot studies based on the authors' knowledge of the targeted area. To select the pilot studies, the authors informally screened the available literature on TSN-5G integration, and selected the followed pilot studies that have a considerable impact on the research area.

 "5G Industrial Networks With CoMP for URLLC and Time Sensitive Network Architecture" (2019) [23].

Table 1

Electronic databases and indexing systems considered in this search.

	0	
Name	Туре	URL
IEEE Xplore Digital Library	Electronic database	www.ieeexplore.ieee.org
ACM Digital Library	Electronic database	www.dl.acm.org
SCOPUS	Indexing system	www.scopus.com
Web of Science	Indexing system	www.webofknowledge.com

- "Extending Accurate Time Distribution and Timeliness Capabilities Over the Air to Enable Future Wireless Industrial Automation Systems" (2019) [24].
- "A Look Inside 5G Standards to Support Time Synchronization for Smart Manufacturing" (2020) [25].

We use these works to formulate the search string and to identify some of the parameters of our data extraction form. Moreover, we perform a manual search in the so-called grey literature (e.g., web pages, forums, etc.) on the Google search engine, to identify any white paper that provides relevant information for our study. Two parallel activities were carried out: the *review of the peer-reviewed literature*, and the *review of the grey literature*.

2.2.1. Automatic search

To identify the relevant studies, we conduct an automatic search on four of the largest and most complete databases in computer science, computer engineering, software engineering, and systems engineering: IEEE Xplore Digital Library, ACM Digital Library, Scopus, and Web of Science as shown in Table 1. The selection of these electronic databases and indexing systems was motivated by their high accessibility and the fact that they export search results to well-defined and computationamenable formats. Furthermore, one of the strongest points is the fact that these databases are recognized as being effective means to conduct systematic literature reviews [27].

We define the search string based on the keywords extracted from the research questions and the pilot studies. We use the same search string in all databases and the process we adapted is based on the search fields required by the digital libraries: title, abstract, and keywords. Our search string is as follows:



Fig. 2. Search and selection process.

The search results include 141 publications in IEEE Xplore Digital Library, 5 in ACM Digital Library, 179 in Scopus, and 115 in Web of Science.

2.2.2. Impurity and duplicates removal

Since some of the research studies can be indexed in more than one database, we first removed the duplicates. Then we addressed impurities in our searches such as abstracts and tutorials. We performed this process by using a tool called StArt [28] that supports the SLR process. After removing the duplicates, we achieved 189 publications – see Fig. 2.

2.2.3. Grey literature search

To collect the grey literature, we follow the guidelines for including grey literature and conducting multivocal literature reviews in software engineering [29]. According to these guidelines we target Google Search Engine, performing the automatic search with the same string search as before, and a manual search based on our knowledge of the targeted research area. The grey literature gave us 10 more primary studies as in Fig. 2.

2.2.4. Application of selection criteria

In the next step, we apply our inclusion and exclusion criteria to all the publications to identify the primary studies that correspond to the relevant publications for this study. In this step, we classify each publication as "Relevant", "Non-Relevant" and "Not Clear". We perform this classification by reviewing the title, abstract, and keywords of each publication. The publication is considered relevant if it addresses the goal and purpose of this study. If the publication is out of the scope of this study, then it is considered non-relevant. Otherwise, if we cannot classify the publication as relevant or non-relevant based on the abstract, title, and keywords, it is set as not clear. In this case, we perform full-text skimming of the publication with the aim of classifying it as relevant or non-relevant. The selection criteria are as follows.

Inclusion criteria for Peer-reviewed literature

- (ICP1) The research study addresses the integration of TSN and 5G.
- (ICP2) The research study is written in English.
- (ICP3) The research study is a peer-reviewed publication, i.e., published in a peer-reviewed journal, workshop, conference, or book.
- (ICP4) The research study is available as full-text.

Inclusion criteria for Grey literature

- (ICG1) Web page or white paper reporting on an integration technique/framework/architecture for TSN and 5G.
- (ICG2) Web page or white paper reporting on a formalization and/or implementation of the proposed TSN and 5G integration approach.
- (ICG3) Web page or white paper is in English.
- (ICG4) Web page or white paper is freely accessible.

Exclusion criteria for Peer-reviewed literature

- (ECP1) Secondary and tertiary studies such as systematic literature reviews, surveys, etc.
- (ECP2) Studies in the form of tutorial papers, short papers, editorials, manual, or poster papers because these types of publications do not provide enough information to answer the posed questions.

Exclusion criteria for Grey literature

- (ECG1) Web page or white paper that does not clearly discuss integration of TSN and 5G.
- (ECG2) Videos, webinars, books, etc. since they are too timeconsuming to be considered for this study.

Even though the secondary/tertiary and other studies are excluded from the search pool, we consider them in identifying any important issues to be considered in our study and for providing a summary of what is already known on TSN-5G integration.

After removing the duplicates and applying the inclusion/exclusion criteria to both the peer-reviewed literature and the grey literature, we managed to get a set of 81 primary studies. The majority of primary studies were excluded due to the removal of impurities and duplicates (251 studies). 91 primary studies were excluded due to not addressing the goal and purpose of this study (not fulfilling ICP1). Moreover, 1 primary study was excluded as it was not written in English (not fulfilling ICP2), and 8 primary studies were excluded as they were not available as full text (not fulfilling ICP4). In addition, 33 primary



Fig. 3. The top-level categories in the data extraction form.

studies were excluded due to being secondary and tertiary studies (ECP1), and 6 primary studies were excluded due to being editorial papers, or posters (ECP2). Note that some of the studies were excluded as they did not fulfill (or fulfilled) more then 1 inclusion (exclusion) criteria.

2.2.5. Snowballing

To mitigate any potential bias regarding the construct validity of our study, we perform a closed recursive backward and forward snowballing activity² [30]. The starting set of our search is the set of primary studies presented earlier and the pool of selected studies after the second string search as shown in Fig. 2. From the recursive backward and forward snowballing, we found two more relevant studies. The full text of one of these studies is not available which violates the inclusion criteria ICG4. Therefore, we remove this study and add the remaining study to the pool of primary studies, which makes the total number of primary studies equal to 82.

The number of relevant studies is low as we expected because the targeted research area is still in its infancy. Note that the integration of TSN and 5G technologies is discussed for the first time in the specification of 5G in 2018. Thus, the research on the integration of TSN and 5G technologies has not gained maturity.

3. Data extraction

In this phase, we create a data extraction form (classification form) that is used to extract the required data from the primary studies. We follow a systematic process based on keywording for defining the parameters in the top levels of the data extraction form in Fig. 3. Furthermore, we use keywording to extract data from the primary studies accordingly.

The goal of the keywording is to effectively develop an extraction form that can fit existing studies while taking their characteristics into account [26]. We collect the keywords and concepts by studying the full text in the primary studies. After collecting the keywords and concepts, we perform a clustering operation to organize them according to the identified categories. The clustering operation is similar to the sorting phase of the grounded theory methodology [31]. During this phase, we collect any additional information that was marked relevant but did not fit within the data extraction form. We refine the data extraction form after the revision of the collected additional information if needed. The previously analyzed primary studies are re-analyzed according to the refined data extraction form. The process is completed only when all the primary studies are analyzed. The final set of analyzed primary studies includes 82 publications.³ We provide a detailed overview of the data extraction form in the following subsections.

3.1. Contribution

This top-level category captures the research contributions concerning TSN-5G integration in the primary studies. Fig. 4 depicts the internal hierarchy of the research contribution category. It consists of three sub-categories: (i) technical contribution, (ii) contribution type, and (iii) maturity level of the contribution. These categories are described as follows.

3.1.1. Technical contribution

This category describes what technical contribution is provided by a primary study. It can be further classified into resource management, flow management, time synchronization, and integration architecture.

The resource management category refers to how the TSN-5G integrated system's resources, including configuration models and scheduling are managed in a primary study [32]. Note that we focus on the data extraction corresponding to the resource management in integrated TSN and 5G networks. We categorize the configuration models according to the IEEE 802.1QCC (2018) [33] as follows:

- *Fully Distributed model:* All the end stations communicate their requirements directly using the Stream Reservation Protocol (SRP) [33]. SRP is a standard that provides mechanisms to reserve bandwidth per queue on the path of a frame. In SRP, a sender can request a reservation for a certain amount of bandwidth in order to transmit a data stream. The reservation request is sent to a reservation server, which determines whether the requested amount of bandwidth is available and, if yes, it allocates it to the sender. The sender can then transmit the data stream using the reserved bandwidth part. Some potential research directions on SRP may be security enhancement, scalability, integration with other protocols, etc.
- Centralized network/distributed user model: There exists a centralized entity called Centralized Network Configuration (CNC) that gathers the information about bridge (TSN switch) capabilities and uses a remote management protocol to perform functions like scheduling, resource reservation and other types of configuration.
- *Fully Centralized model:* This model contains, in addition to the CNC, an entity called the Centralized User Configuration (CUC) that collects all user requirements and interacts with the CNC to decide the topology and scheduling of the network.

The flow management category refers to data models or protocols that enables users or operators to dynamically discover, configure, monitor, and report the bridge and end station capabilities [7]. Among several models and languages, the prominent one in the context of TSN and 5G is the YANG model that is used to model configuration data, state data, Remote Procedure Calls (RPCs), and notifications for network management protocols [34]. The YANG model is a formal contract language used for networking, and widely adopted in industries. This is the main motivation why the TSN Task Group decided to establish IEEE 802.1QCP standard to support YANG data modeling. The YANG model is used by the widely accepted protocols, such as NETCONF and RESTCONF, to simplify network configuration, as described below.

- NETCONF is a network management protocol which provides mechanisms to install, manipulate, and delete the configuration of network devices [35]. It is used by the centralized entity of a TSN configuration model (CNC) to configure the switches following a client–server model [36].
- RESTCONF is a network management protocol used to provide the Create, Read, Update, Delete (CRUD) operations on a conceptual data store containing YANG-defined data [37]. It provides an interface to NETCONF data stores leveraging the HTTP methods. In the fully centralized TSN configuration model, it appears as an interface between the CNC and CUC entities.

² Backward snowballing refers to scouring the references sections of papers that are already included in the review, while forward snowballing refers to tracking articles that had subsequently cited any of the papers included in the review.

³ The final set of primary studies corresponds to studies published until **January 21, 2023**, which corresponds to the last day of our search.



Fig. 4. Data extraction, considering all the main topics in terms of technical issues, types of contributions, and maturity level.

In addition to the above well-known models, Stream Reservation Protocol (SRP) [33] and Link-Local Reservation Protocol (LLRP) [38] are the flow management protocols used solely in TSN and wireless networks, respectively. They help in improving the performance and reliability of the network by ensuring necessary resources to critical devices and traffic.

Time synchronization category refers to the synchronization of TSN and 5G to achieve a whole unified and converged network. Synchronization may vary depending on the time synchronization approach and/or the source of synchronization. We also consider the studies that do not address the time synchronization approach. We categorize time synchronization approaches according to TR 23.734 [39] as follows:

- *Boundary clock approach:* The 5G Radio Access Network (RAN) has a direct connection to the TSN master clock and the timing information is provided to User Equipment (UE) via the 5G broadcast channels.
- Transparent clock approach: This approach uses the generalized Precision Time Protocol (PTP) [40] messages to achieve synchronization. The gPTP is a network protocol used to synchronize the distributed clocks within a communication network. The synchronization of clocks between network devices is achieved by passing relevant time event messages [41].

Even though the 3GPP working group addresses synchronization accuracy in the range of hundreds of nanoseconds, this accuracy still depends on the suggested approach. The integration architecture category refers to the proposed TSN-5G architecture. This architecture can either conform or not conform to the 3GPP Releases. In the latter case, the architecture is often designed according to the authors' knowledge of the targeted area. Most of the works follow the 3GPP Releases, which are developed in a backward compatible manner by the 3GPP working group. Release 17 is still under development but there are several researchers that already refer to this release in their works.

In addition, there are different properties of the architectures that can be addressed. We identify the following core properties:

- *Programmability:* using different algorithms to dynamically reprogram the nodes. The current effort on network programmability is mostly centered around the separation of the data and control planes [42,43].
- *Interoperability:* different networks can communicate easily without the need for additional tools or interfaces. This concept is not only related to communication but also to the specification and implementation of an application [44]. Interoperability can be categorized with respect to the resources, protocols, services, and timeliness of the communication. Further definitions of interoperability are presented in [45,46].
- *Dependability:* the ability of a system to provide services that can be trusted within a time period. It includes the system's availability, reliability, maintainability, maintenance support, and performance and in some cases, it may include durability, safety and security [47,48].
- Predictability: is related to proving, demonstrating, or verifying the fulfillment of the system's timing requirements. In the artificial intelligence community, predictability is also related to the support for mechanisms that predict beforehand the future state of the system [49,50].
- Mobility: refers to network mobility. Based on some measurement reports, the network may possibly move (i.e., handover [51]) the mobile terminal connection from the serving cell to that neighbor cell, so the mobile terminal will get better radio conditions [52].
- *Heterogeneity*: in the sense of a network containing different types of nodes as in [53]. Software, hardware, and technology variation between mobile devices cause heterogeneity [54].
- *Backward compatibility:* is the ability of a network to be compatible with earlier versions, meaning that all the previous features will be valid in the new version [55]. For example, 5G devices are able to operate on earlier-generation networks (4G, 3G, etc.).
- Redundancy: is the duplication of network instances such as devices or lines of communication to increase the system's reliability and to reduce the risk of failures. It is one of the mechanisms to provide reliable data transfers [56,57].

3.1.2. Contribution type

Another sub-category of the contribution is the contribution type. It consists of (i) method, technique, or approach, (ii) model, architecture, or framework, and (iii) tool. A tool can be:

- *Proprietary:* A tool that is not commercially available but there is a party that has the right to grant a license for using it.
- Freely available: Open for all users.
- *Commercial:* This tool has a commercial purpose but licence might be needed in order to use it or it can be open-source depending on developers' decision.

3.1.3. Maturity level

Maturity level of a study is described based on maturity classification according to the Redwine–Riddle maturity model [58]. Accordingly, Score 1 means "not mature at all", so the study only presents the basic ideas but there is no proof of concept. Score 2 means "somewhat mature", the study provides a proof of concept, and the contribution is



Fig. 5. Data extraction form, considering different purposes of each primary study.

also demonstrated on use cases. The highest is a score 3, which means "mature", the study includes the proof of concept and the usability presented on one or more use cases. Furthermore, there is evidence of the usage of the contribution by the research community. Score 0 means "inconclusive", the contribution of the study cannot be classified as any of the above ones.

3.2. Purpose

The second top-level category in the data extraction form is denoted by "purpose". This category classifies the primary studies according to the purpose of the contribution. We identify three main purposes of the contributions in the area: (i) to **optimize** specific aspects of a technique, (ii) to **design** a new technique or an architecture, and (iii) to **evaluate** a current technique or architecture as depicted in Fig. 5.

3.2.1. Optimization

The optimization refers to the end-to-end schedule optimization [59] or cost optimization. End-to-end schedule optimization considers the following parameters:

- *Latency:* measures the time it takes for data to propagate from its source node to its destination node via the network.
- Jitter: a variation in delay usually caused by network congestion.
 Resource efficiency: addressed in some of the primary studies
- which aim to achieve an improved schedule of the network.
- Supported number of time-sensitive applications: affects the timing requirements and overall performance of the network.

Cost is a broad aspect of optimization, which can be further categorized as follows:

- · Power Consumption
- · Bandwidth
- · Product Costs

- · Complexity and implementation costs
- Infrastructure cost
- · Cost of additional network resources

3.2.2. Design

The purpose of a primary study could be to develop a technique or a method for designing an architecture or a prototype of a TSN-5G integrated system. The designed prototype could represent, for example, a hardware and/or software for the network interface that supports connectivity, synchronization, scheduling, and analysis of TSN-5G integration.

3.2.3. Evaluation

The purpose of a primary study could also be to evaluate a technique or a method for TSN-5G integration using formal and/or empirical analysis. The analysis can be performed on a real network, simulated network, or hybrid network. Furthermore, the workloads used for the evaluation could be (i) synthetically generated, (ii) acquired from real network traffic, or (iii) based on the traces generated by running real traffic.

3.3. Domain

The third top-level category in the data extraction represents the application domain of the research contribution presented in each primary study. The proposed contribution in a primary study could be applicable generally or proposed for a specific domain or a particular segment of the industry, e.g., industrial automation [60,61], automotive [62–64], railways, avionics, etc.

3.4. Research type

The general taxonomy of classifying research studies based on their research type is summarized in [65]. To classify the research type in the primary studies, we use a reduced classification of the general taxonomy as presented in [66]. We classify the primary studies using the following research types.

- I. *Solution Proposal* This type of research proposes a new technique or an extension of an existing technique and discusses its relevance. Small examples, a sound argument or other means may be used to provide a proof of concept of the proposed/ extended technique.
- II. Validation Research This type of research aims at investigating some properties of a solution proposal and presenting a novel technique or method that is not yet implemented in practice. The investigation is conducted systematically utilizing various activities, including the development of prototypes, performing simulations, conducting experiments, performing mathematical analysis, and mathematically proving the properties.
- III. Conceptual (Philosophical) Proposal This type of research proposes a new conceptual framework. Furthermore, using the framework provides a new way of looking at the problem at hand.
- IV. Evaluation research This type of research evaluates an implemented solution in practice (real environment). The evaluation is often conducted by means of case studies, field studies, or field experiments.
- V. *Experience paper* This type of research presents the lessons learned by the researchers from their own experience of solving the research problem. These kinds of studies are performed by researchers who have used some tools in practice or by industry practitioners who report their experience of studying the problem in industrial settings.

Table 2

Categorization and tabulation of primary studies based on the technical contributions.

Technical category	Reference to the paper
Resource management	[23-25,60-62,68-111]
Flow management	[23,24,60,61,68–71,73–76,78,79,81–85,96,98,99, 107,108,110,112–114]
Time synchronization	[23–25,61,62,68–71,74–76,78,79,81,83–85,87–91, 93,94,97,98,100,102,104–108,111–113,115–137]
Integration architecture	[23-25,60-62,68-143]

4. Results: Vertical analysis

We follow the data analysis guidelines provided by Cruzes and Dyba [67] to perform the vertical and horizontal analyses of the extracted data that we gathered in this study. The vertical analysis, performed in this section, is used to find information about each of the categories identified in the data extraction form. As the first step, we analyze each primary study specifying the parameters that the extraction form requests, and then we analyze the entire pool of primary studies to find any potential gap in the existing research. Note that we also provide summary tables (Tables 2–13) where each primary study is connected to each specific category and its characterizing value(s). The references underlined are common between two different categories.

On the other hand, horizontal analysis (presented in the next section) is used to identify the possible relations between two different categories, showing the trends and potential gaps through contingency tables.

This section provides an analysis of each of the categories of our data extraction form and answers our first and second research questions. We access trends and venues of primary studies over time. Furthermore, we investigate the existing approaches and techniques for TSN-5G integration in terms of their technical characteristics. The selected set of primary studies is presented in Table 2. This table categorizes the primary studies based on the technical contributions specified in Fig. 4. We choose this special category as it includes all the set of primary studies considering our goal to focus particularly on TSN-5G integration.

In the following subsections, we provide an answer to our first research question: *RQ1: What are the technical characteristics of TSN and 5G integration?*

4.1. Analysis based on technical contributions

In this subsection, we investigate the set of primary studies based on their technical contributions to the research area. Based on our classification scheme, we analyze the technical contributions of the studies based on the aspects they address, which can be resource management, flow management, time synchronization, and integration architecture, as shown in Fig. 6. A summary table of the primary studies belonging to each technical category was presented earlier in Table 2.

We notice that 73% of the primary studies address the time synchronization, which is actually a significant challenge for the integration of TSN and 5G technologies. 60 primary studies focus on time synchronization of TSN and 5G. The majority of the primary studies propose a synchronized approach with an accuracy starting from the levels of hundred nanoseconds to one microsecond. There are six primary studies [81,85,87,98,103,121] that do not consider time synchronization, but rather propose non-synchronized solutions for TSN and 5G integration. Usually, the choice of synchronized or non-synchronized solution depends on the specific requirements of the system. Clock synchronization is a critical aspect in real-time systems, where the timing of events and the order in which they occur can have significant consequences.



Fig. 6. Number of primary studies providing various technical contributions for TSN-5G integration.

Table 3 A summary of primary studies according to the approach they use to achieve the time synchronization.

Time synchronization approach	Reference to the paper
Transparent clock approach	[24,25,62,71,79,83,88,91,93,94,97,100,102,104, 106,111,113,115–118,120,122–124,126–129, 131–133,135,137]
Boundary clock approach	[23,125]

The majority of the primary studies (34) that address timesynchronization use the transparent clock approach that utilizes the Precision Time Protocol (PTP). Whereas, there are only two primary studies that use the boundary approach as shown in Table 3.

The transparent clock approach is usually used in high-precision applications, to maintain a very accurate view of the current time because it takes into account the delays introduced by the network itself. On the other hand, the boundary clock approach is simpler to implement and maintain, but it is less accurate as it depends on the accuracy of the clocks at the boundary of each subnetwork, which may not be as precise as more central clocks. Note that both approaches are proposed by the 3GPP standards.

There are 50 primary studies (61% of the total) that provide technical contributions in the context of resource management in TSN-5G integrated systems. Among these studies, 31 use the fully centralized configuration model proposed by the IEEE 802.1 QCC standard (Table 4). One advantage of the centralized model compared to the distributed one is that the centralized model can be easily incorporated into the specifications provided by 3GPP to manage and control the TSN-5G network.

On the other hand 23 primary studies focus on providing different scheduling techniques for TSN-5G network. Among the scheduling algorithms semi-persistent scheduling (SPS) is used by the majority of the papers, as shown in Table 5. In SPS, a dedicated channel is reserved for a specific user for a predetermined period of time. SPS is a useful technique for supporting the transmission of periodic data with low latency and high reliability in wireless communication systems, however it may not be suitable for traffic that is highly variable or unpredictable in nature, as it may require a more flexible scheduling approach.

Table 4

F	summary	ot	primary	studies	according	to	the	network	configurat	lon	model.	

Configuration model	Reference to the paper
Fully distributed model	[25]
Centralized network/distributed user model	[23]
Fully centralized model	[24,25,60–62,68,71,72,74,76,78–83,86,88,89,91– 93,96,100,101,103,106–109,111]
Others	[69,73,85]

Table 5

A summary of primary studies according to the proposed scheduling technique.

Scheduling technique	Reference to the paper
Semi-persistent scheduling	[23,24,60,76,77,99,100]
Configured grant scheduling	[60,71,100]
Dynamic scheduling	[68]
Window-based scheduling	[86]
Others	[61,72,74,75,84,85,95,102,106,108–111,118]



Fig. 7. Number of primary studies conforming to various releases of the 3GPP specification.

The analysis of the primary studies also reveals that 74% of these studies (representing % of all studies) contribute towards the integration architectures for TSN and 5G that conform to the 3GPP specifications (Release 15, 16, 17 or others as shown in Fig. 7). Whereas, the TSN-5G integration architectures addressed in the remaining primary studies (26% of the total number of primary studies) do not comply with any of the 3GPP Releases [68–70,72,80,84,85,95–99,110,121,124, 130,132–134,136,142].

In addition, the number of primary studies addressing each property of the TSN-5G integration architecture is depicted in Fig. 8, while a summary table of the primary studies belonging to each integration architecture property is presented in Table 6. Overall, dependability is an essential requirements for TSN-5G network as it enables the transmission of time-critical data with low latency and high reliability. TSN networks need to be secure in order to prevent unauthorized access to the critical data. Ensuring the security of TSN in a 5G environment is one of the major challenges, as it requires the integration of multiple security technologies. On the other hand, interoperability is also a challenge, as TSN and 5G components may use different protocols and technologies. It should be noted that one of the challenges of TSN-5G integration comes from the standardization. TSN is still an evolving standard, and this can lead to challenges in terms of interoperability and compatibility as there might be ongoing efforts to standardize TSN-5G integration.



Fig. 8. Number of primary studies addressing various properties of integration architectures for TSN and 5G.

Table 6

A summary of primary studies according to the properties of integration architectures for TSN and 5G. Note that the references underlined are common between two different categories.

Architecture's properties	Reference to the paper
Programmability	[24], [60], [61], [68], [69], [72], [73], [85], [86], [87], [92], [93], [103], [132], [110], [108],
Interoperability	$\begin{array}{c} [23], [24], [60], [61], [68], [72], [74], [81], [82], \\ [83], [88], [89], [92], [95], [96], [100], [101], \\ [102], [103], [104], [119], [120], [121], [138], \\ [141], [106], [114], [107], [108], [111], [142], \\ \end{array}$
Dependability	
Predictability	[23], [24], [60], [61], [69], [76], [80], [102], [115], [116], [117], [122], [139], [108], [136], [137]
Mobility	[24], [70], [62], [76], [78], [79], [113], [139]
Heterogeneity	$\underbrace{[68], [70], [72], [73], [74], [82], [85], [84], [92],}_{[19], [101], [105], [139], [114], [108]}$
Backward compatibility	<u>[82]</u> , <u>[87]</u> , <u>[88]</u> , <u>[93]</u> , <u>[103]</u> , <u>[116]</u> , <u>[121]</u>
Redundancy	<u>[23], [24], [25], [61], [68], [71], [78], [81], [84],</u> <u>[118], [119], [134],</u>
Others	[25], [75], [77], [81], [91], [93], [94], [97], [98], [99], [123], [124], [125], [126], [127], [128], [129], [130], [131], [132], [133], [141], [135]

4.2. Analysis based on the contribution type

In this subsection, we discuss the distribution of the primary studies considering their type of contribution as presented earlier in Section 3.1.2. Fig. 9 presents the number of primary studies addressing each contribution type. It can be observed that 37 primary studies focus on providing a method, technique, or approach for TSN-5G integration. Similarly, a set of 27 primary studies provide a model, architecture, or framework for TSN-5G integration.

It is interesting to note that none of the primary studies have provided a tool as a contribution (Table 7). Often, tools are the means



Fig. 9. Number of primary studies providing the type of contribution with regards to TSN-5G integration.

Table 7

A summary of primary studies according to the type of contribution with regard to TSN-5G integration.

Type of contribution	Reference to the paper
Method	[25,62,71,73,74,80,85,86,92–94,97,100,101,103,104,109, 110,112,114–118,120–124,126–128,130,135,137,140,142]
Model	[23,60,68–70,72,76,78,79,81,82,84,91,95,96,98,99,102, 108,113,119,125,132,133,136,138,143]
Tool	-
Others	[24,61,75,77,83,87–90,105–107,111,129,131,134,139, 141]

to transfer scientific results to the industry. This indicates that there is an urgent need for provisioning of tools that incorporate the existing techniques or new techniques for TSN-5G integration. The remaining 18 primary studies provide other contributions, e.g., to provide options for TSN and 5G, to investigate the impact and derive requirements on the end-to-end system, to present the key technologies, challenges, and research directions.



Fig. 10. Distribution of primary studies according to research type classification.

Table 8

A summary of primary studies according to research type classification.

Research type	Reference to the paper
Solution proposal	[24,25,60,62,68,70–73,75,76,79,83,88–90,93,94,96,100,106, 108,112,116,120,121,124,129,130,135,136,138,139,143]
Validation research	[23,61,69,74,77,78,80-82,84-87,92,95,97-99,101-104,109- 111,113,115,117-119,122,123,125-128,132-134,137,140, 142]
Conceptual proposal	[91,105,107,131,141]
Evaluation research	-
Experience paper	-

4.3. Analysis based on the research type

There are 5 research types that we presented as part of the data extraction form: (i) solution proposal, (ii) validation research, (iii) conceptual (philosophical) proposal, (iv) evaluation research, and (v) experience paper. We identify that 42 primary studies provide validation research for TSN-5G integration. Similarly, 35 primary studies present solution proposals for the integration of TSN and 5G as shown in Fig. 10. These studies develop proof-of-concept prototypes and perform simulations and mathematical analysis using the prototypes. Only 5 primary studies provide conceptual proposal for TSN-5G integration. It is interesting to note that none of the primary studies provide evaluation research or experience papers (Table 8). This analysis identifies a gap in the research types in this area, which is also an indicator of the immature nature of TSN-5G research.

4.4. Analysis based on the maturity level

In this subsection, we analyze the primary studies with respect to the maturity levels according to the Redwine–Riddle maturity model [58], also described in Section 3.1.3. According to this model, we classify the primary studies using one of the scores: 0, 1, 2, or 3 representing "inconclusive", "not mature at all", "somewhat mature" or "mature" respectively. It can be observed in Fig. 11 that 55% of all primary studies are somewhat mature as the technical contributions in these studies have been demonstrated in use cases involving TSN-5G integration. Whereas, 43% of all primary studies are not mature at all as they only present basic ideas without providing any proof of concept for TSN-5G integration. It is interesting to note that only two of the primary studies are mature according to the Redwine–Riddle maturity model.



Fig. 11. Distribution of primary studies according to the maturity level of their contributions with regard to TSN-5G integration.



Fig. 12. Number of primary studies according to the purpose of contribution in the context of TSN-5G integration.

4.5. Analysis based on the purpose of contributions

In this subsection, we analyze each primary study based on the purpose of the contribution as discussed in Section 3.2. Fig. 12 presents the number of primary studies that aim at evaluating, designing, or optimizing some techniques for TSN and 5G integration. We notice that a large majority of the primary studies (70) focus on optimizing the TSN-5G integrated systems. Furthermore, 68 primary studies focus on design, whereas 57 primary studies provide an evaluation of the techniques for TSN-5G systems. Note that several primary studies have more than one purpose in the proposed contributions, e.g., to design and evaluate, to design and optimize, or to optimize and evaluate.

4.5.1. Purpose of contribution – optimization

If the purpose of contribution in a primary study is to perform optimization, then we identified two sub-categories in our data-extraction form: end-to-end schedule optimization and cost optimization. Fig. 13 shows that 74% of the primary studies aim at optimizing latency, which is a key quality-of-service attribute in many applications in various domains, e.g., Industrial Internet of Things (IIoT). Achieving TSN low latencies in a 5G network may be challenging due to the high traffic volumes and potential interference from other devices.



End-to-end schedule optimization parameters

Fig. 13. Number of primary studies addressing various attributes for end-to-end schedule optimization in TSN-5G systems.

Table 9

A summary of primary studies according to various attributes for end-to-end schedule optimization in TSN-5G systems.

Scheduling parameter	Reference to the paper					
Latency						
Jitter						
Resource efficiency						
Supported number of additional resources	[61], [74], [77], [87], [115], [121]					

The second most commonly addressed attributes are jitter, and resource efficiency, which are subject to optimization by 24% and 23% of the primary studies respectively. TSN can contribute to resource efficiency in 5G networks by its ability to support fine-grained, deterministic scheduling of network resources. This allows TSN to allocate network resources in a precise and predictable manner, optimizing the use of network resources and reducing congestion. Furthermore, 7% of the primary studies focus on the optimization of additional resources like the supported number of time-sensitive and/or background applications. Overall, 5G networks are designed to support a wide range of applications and services with different set of requirements, including high data rates, low latency, and high reliability.

Many primary studies also address optimization of various types of costs in the context of TSN-5G integration as shown in Fig. 14. Optimization of bandwidth cost in TSN-5G integration is the most studied topic, which is addressed by 14 primary studies. Furthermore, optimization of the costs of "Complexity and implementation", "infrastructure", and "cost of additional network resources" in TSN-5G integrated systems are addressed by 6 primary studies each. Similarly, a few studies have also addressed optimization of product costs, power consumption, and other resources (e.g., flow acceptance ratio) in TSN-5G integrated

Table 10

A summary of	primary	studies	according to	various	types	of cost	optimization	in	TSN-5G
systems.									

Type of cost	Reference to the paper
Power consumption	[24], [68], <u>[69]</u> , <u>[139]</u>
Bandwidth	[23], [69], [70], [73], [82], [86], [98], [118], [121], [138], [114], [109], [142], [143]
Product cost	<u>[98]</u> , <u>[139]</u>
Complexity and implementation	[61], [62], <u>[73]</u> , [78], <u>[82]</u> , [112]
Infrastructure	[69], [72], [82], [105], [115], [108]
Cost of additional network resources	<u>[73]</u> , [84], [85], <u>[86]</u> , [92], [110]
Others	[81], [123]

Table 11

A summary of primary studies according to the type of design of TSN-5G integrated systems.

Type of design	Reference to the paper
Software design	[23], [24], [25], [61], [62], [68], [71], [74], [75], [76], [77], [79], [80], [81], [82], [83], [84], [85], [86], [87], [90], [92], [93], [94], [95], [96], [97], [98], [99], [100], [101], [103], [112], [115], [116], [117], [118], [120], [122], [123], [124], [125], [126], [127], [128], [130], [138], [139], [141], [132], [133], [114], [107], [108], [109], [110], [135], [136], [142]
Hardware design	[60], [69], [87], [102], <u>[103]</u> , [119], [121], <u>[130]</u> , [131], <u>[138]</u> , [134], [111]

systems as shown in Fig. 14. Implementing TSN-5G networks can be expensive, as it requires the deployment of additional hardware and software. Note that some primary studies address optimization of more than one type of costs (Table 10).

4.5.2. Purpose of contribution - design

We identified that 72% (59 studies) of all the primary studies provide software design for the technique to integrate the TSN and 5G technologies. Furthermore, we observe that only 12 primary studies (15% of the total) present hardware design for the TSN and 5G integration. We also note that only three primary studies [103,130,138] (Table 11) addresses both the software and hardware design of the suggested approach for TSN-5G integration. TSN-5G hardware devices need to have certain capabilities including support for IEEE 802.1 standards, clock synchronization, QoS mechanisms such as packet classification, scheduling, and traffic shaping, etc. Overall, the hardware design of TSN-enabled devices in a 5G network will depend on the specific requirements of the application, however it may take some time before TSN-5G devices become widely available on the market.

4.5.3. Purpose of contribution - evaluation

We observe that a large majority of the primary studies (49 studies) use empirical evaluation for the proposed technical contributions. On the other hand, 20 primary studies use formal analysis to evaluate the proposed technique(s) for TSN-5G integration. Fig. 15 shows the number of primary studies per evaluation setup. The empirical analysis is mostly used in our set of primary studies (Table 12), assuming mostly a simulated network in N3, OMNeT++, or other simulation tools using synthetically generated workloads. These tools include a TSN module, as well as a 5G module to simulate TSN-5G networks. The choice of the simulation tool will depend on the specific requirements and goals of the simulation.

The majority of the primary studies (29 studies) use simulated networks for their empirical evaluation, while 19 studies use a real network to perform the evaluation of their TSN-5G suggested technique/approach as shown in Fig. 15(a). We note that there are no



Fig. 14. Number of primary studies addressing various types of cost optimization in TSN-5G integrated systems.



(a) Distribution of primary studies based on the type of empirical evaluation.



(b) Distribution of primary studies based on the type of workload used for the empirical evaluation set up.

Fig. 15. Empirical evaluation strategies used by the primary studies.

publications of type hybrid, considering an integration between real hardware and simulated network components. Another part of the empirical evaluation is the type of workload used in the evaluation set up as shown in Fig. 15(b). We notice that the majority of primary studies (24 studies) use a workload of type synthetic, while there are six primary studies having actual real traffic on their evaluation set up, and two primary study using traces from real traffic. Considering the immature nature of TSN-5G networks, it is hard for the researchers to test and evaluate their approaches using real traffic from an established TSN-5G network.



Fig. 16. Publication trend in the research on TSN-5G integration.

4.6. Analysis based on the domain

The domain where a technique or framework is applied can be generic, or specific depending on the focused area. As shown in Table 13, 55 primary studies (67% of the total) provide domain-agnostic techniques for the integration of TSN and 5G technologies. There are 22 primary studies (24% of the total) that are focused on the industrial automation domain. There are only three primary studies that focus on the automotive domain. The research on TSN-5G integration is still in its early stages which explains the large number of primary studies that have no concrete application domain in focus.

4.7. Publication trends

This subsection presents the publication trends based on the extracted data from the primary studies, thereby addressing the second research question: *RQ2: What are the publication trends of TSN and 5G integration?* To answer this question, we extract the year of publication and venue from each primary study.

Publication year. As shown in Fig. 16, publications in this research area commenced in 2018 after the initial delivery of 3GPP Release 15 in late 2017, which was the first full set of 5G standards. This shows that the research area is still in its infancy. The number of primary studies started to increase in 2019, 2020, and 2021 with 10, 24 and 26 publications respectively. The results also show that the interest of researchers in the area is continuously growing.

Publication venues. We analyze the primary studies based on the publication type, which can be a journal, a conference, a workshop, or a book chapter. The results reveal that most of the primary studies (47 out of 82) are published in conferences. 24 primary studies are

Table 12

A summary of primary studies according to the empirical evaluation strategies.

Empirical evaluation		Reference to the paper
Type of evaluation	Real network	[23], [60], [72], [82], [87], [102], [103], [104] [116], [118], [119], [121], [132], [133], [138], [134], [111], [136], [137]
	Simulated network	[23], [61], [69], [70], [77], [80], [81], [84], [85], [86], [92], [95], [96], [97], [98], [99], [101], [113], [115], [117], [120], [106], [122], [127], [140], [108] [109], [110], [142]
	Hybrid	-
Type of workload	Synthetic	
	Actual real traffic	[23], [102], [103], [104], [138], [111]
	Traces from real traffic	<u>[98]</u> , <u>[116]</u>

Table 13

Number of primary studies per each type of domain.

Type of domain	Number of primary studies	Reference to paper	
Generic	55	[23,25,68-75,77,79,81,82,85-88,91-97,99,100,103-105,107,110,112,113,115,117,118,121-133,135-137,140,141]	
Industrial automation	22	[24,60,61,76,78,80,83,89,90,98,102,106,108,109,111,116,120,130,134,138,139,143]	
Automotive	3	[62,101,142]	
Smart cities	1	[119]	
Medical	1	[84]	

Table 14

Publication venues with at least 2 primary studies.

Venue name	Acronym	Туре	#papers
IEEE International Workshop on Factory Communication Systems	WFCS	С	6
IEEE International Conference on Emerging Technologies and Factory Automation	ETFA	С	6
IEEE Communications Standards Magazine	-	J	5
IEEE Network	-	J	3
IEEE Access	-	J	3
IEEE International Conference on Edge Computing	EDGE	С	2
International Conference on Information and Communication Technology Convergence	ICTC	С	2
IEEE International Conference on Communications	ICC	С	2

published in journals with at least 3 journal publications per year since 2019, 1 primary study is a book chapter, while the remaining primary studies are part of our grey literature search consisting of white papers, web pages, or forums.

Another classification is based on the targeted publication venues. In Table 14, we list the publication venues where at least two primary studies have been published. The IEEE International Workshop on Factory Communication Systems (WFCS) and the IEEE International Conference on Emerging Technologies and Factory Automation (ETFA) have published the highest number of primary studies to date.

5. Results: Horizontal analysis

In this section, we investigate the possible relations that might exist between different categories of the data extracted from the primary studies. The purpose of this analysis is to highlight the main focus and identify the potential gaps in the existing research on the integration of TSN and 5G technologies. This analysis aims to provide an answer to the third research question, *RQ3: What are the limitations of TSN-5G integration?*

We analyze the relationship between two different categories using bubble plots in which the size of the bubble corresponds to the number of primary studies addressing the pair of categories intersecting each other. The first bubble plot, depicted in Fig. 17, shows the relationship between the technical contribution classification (along the vertical axis) and the research type classification (along the horizontal axis). As can be noticed from the plot, the majority of the existing research in the area is focused on presenting solution proposals and validation research for TSN-5G integration techniques. 35 primary studies provide solution proposals for TSN-5G integration architectures. Furthermore, there are 26, 14, and 22 primary studies that provide solution proposals for time synchronization, flow management, and resource management in the context of TSN-5G integration respectively. Similarly, there are 42, 25, 12 and 23 primary studies that present validation research for integration architectures, time synchronization, flow management, and resource management respectively. In addition, there are a few primary studies that present a conceptual proposal for TSN-5G integration. It can be observed in Fig. 17 that there are no evaluation research and experience papers that provide a technical contribution (integration architectures, time synchronization, flow management, and resource management) for TSN-5G integration. This identifies a potential gap in the existing research and provides opportunities for further research on the integration of TSN and 5G technologies.

The relationship between the technical contribution classification and the contribution type classification in the primary studies is illustrated by the bubble chart in Fig. 18. It can be observed from the bubble chart that the majority of primary studies provide a method/technique/ approach or a model/framework/architecture for the integration of TSN and 5G technologies. For instance, there are 37, 26, 7, and 17 primary studies on TSN-5G integration that provide a method, a technique, or an approach for the integration architectures, time synchronization, flow management, and resource management respectively. Similarly, there are 27, 17, 16, and 19 primary studies on the integration of TSN and 5G technologies that provide a model, an architecture



Fig. 17. Relationship between the research type and technical contribution classifications in the primary studies.



Fig. 18. Relationship between the technical contribution and contribution type classifications in the primary studies.

or a framework to support the integration architectures, time synchronization, flow management, and resource management, respectively. Integration architecture is addressed in most of the primary and this can be motivated by the converged nature of a new network which still needs a lot of research and investigation.

The bubble chart in Fig. 18 also indicates that there are a few primary studies that provide other types of contribution with regards to the TSN-5G integration architecture, time synchronization, flow management, and resource management, e.g., solutions for TSN on 5G fronthaul [75]. One major gap that we identify in the existing research

on the TSN-5G integration is that there is no tool support available with any of the proposed techniques. This is evident from zero entries in the "Tools" column in Fig. 18. Note that tools serve as a vehicle to transfer research results from academia to industry. This calls for the research community to develop prototypes of a tool that implements the scientific techniques for the integration of TSN and 5G technologies.

6. Fleiss kappa analysis

Fleiss kappa is a statistical analysis that can help in defining the level of agreement among the researchers during the selection phase of this systematic literature review. We will use it to address the possible threats to validity and to ensure the reliability of our study inclusion decisions.

The idea of such statistical analysis came firstly from Cohen [144] who considered a case when two raters were trying to rate or categorize a set of subjects. He introduced a kappa value to measure the level of agreement between two raters. Fleiss kappa statistical analysis [145] was introduced to eliminate the limitation of only two raters. We use this method in the early phase of the selection of publications to access the reliability of agreement among the researchers performing this study.

To perform the Fleiss Kappa analysis, three researchers were assigned 25 publications each. The publications were selected randomly from the search pool of publications taken after the removal of duplicates and impurities as shown in Fig. 2. The researchers independently categorize those publications as Relevant, Not Relevant, or Not Clear based on title, abstract, keywords, and full-text skimming (if needed). From the Fleiss Kappa point of view, the 25 papers are the subjects that need to be categorized by the 3 raters into 3 categories. After **applying the analysis, we calculated the overall agreement among the researchers on a value of 83% which indicates a strong level of agreement among the raters [146]. Based on this statistical analysis we conclude that the researchers had a strong level of agreement when selecting the relevant papers for the systematic literature review.**

7. Threats to validity

To prove the quality of our study, we discuss in detail the possible threats to validity and how we managed to mitigate them. There are three types of threats we address: external, internal, and construct validity.

7.1. External validity

The external threat to validity deals with the generalizability of the results [147]. A possible threat can be a set of selected studies that cannot fully represent the state of the art on TSN-5G integration. To mitigate this potential threat, we make sure to choose multiple data sources which are four of the largest and most complete databases in computer science, computer engineering, software engineering, and systems engineering: ACM Digital Library, IEEEXplore Digital Library, Scopus and Web of Science. After the automatic search, we implement the recursive backward and forward snowballing strategy to be sure about the coverage of our study.

Considering the pilot studies in our systematic literature review, we apply well-defined and constructed inclusion and exclusion criteria. Excluding the studies which are not written in English can be a possible threat, but considering the fact that English is the de-facto language for all the scientific papers, this threat can be omitted.

7.2. Internal validity

Internal validity refers to the inaccurate settings or variables that may cause a negative impact on the design of our systematic literature review. We mitigated this threat by following well-established guidelines when defining the data extraction form and the process which we follow in this study. Furthermore, we cross-analyzed all the parameters in the data extraction form to identify and solve any potential issues with the consistency of the extracted data.

7.3. Construct validity

Construct validity refers to the representativeness of the selected studies [22]. The recursive backward and forward snowballing makes us confident on the coverage of our study that we did not miss any relevant study.

In the beginning, we collected the research studies using the search string. The definition of the string can be a potential threat to validity. All the researchers involved in this study discussed together every parameter of the search string by following a rigorous process. This minimizes the threat to construct validity. After the automatic search, studies were analyzed by well-documented inclusion and exclusion criteria. To prove the reliability of the review, three of the researchers independently classified a set of common studies and applied the Fleiss kappa statistical analyses [145] as described in Section 5, achieving a kappa value of 83% which means a strong agreement among the researchers.

7.4. Conclusion validity

The conclusion validity refers to the relationship between extracted data and obtained findings [147]. We mitigated this potential threat by systematically documenting by using a well-defined process and by providing a replication package that allows replicating each step of the process. The replication package is freely accessible.⁴ The definition of the data extraction form can be a potential threat to conclusion validity. To mitigate this threat we (i) let the parameters emerge from the pilot studies and refine the parameters throughout the entire data extraction activity, and (ii) make all the researchers actively involved in the definition of the extraction form as well as in the extraction and analysis of the data.

8. Related work

There are a few systematic reviews and surveys conducted in the area of TSN and 5G with a special focus in 3GPP Releases. For example, the study in [148] presents an overview of 3GPP Releases focusing on the extensive enhancements to achieve backward compatibility in the subsequent releases. Similar studies are performed by Jerichow et al. [149], Nwakanma et al. [150], and Atiq et al. [32].

Jerichow et al. [149] present an overview of public networks that are integrated with non-public networks within the scope of the 3GPP Release 16 architecture. Furthermore, they also discuss security concepts of 5G non-public networks. Nwakanma et al. [150] review the implementation possibilities and challenges of 3GPP Release 16 in Industrial Internet of Things and mission-critical communications. Atiq et al. [32] comprehensively analyze the recent standardization efforts and developments in IEEE 802.11 and 5G, and present a set of use cases enabled by wireless TSN. The authors provide insights in wireless TSN considering time synchronization between 802.11 or 5G and TSN devices, techniques to achieve reliability requirements, and mapping QoS profiles with the TSN defined traffic.

Jun et al. [151] perform a detailed survey of 3GPP standardization activities to ensure low latency at the network level. Furthermore, they investigate the time-sensitive communication in Release 16 and 17, including the time synchronization in a TSN-5G architecture that conforms to the two Releases. Wollschlaeger et al. [152] also present

an overview of the 5G evolution among 3GPP Releases starting with Release 15 and concluding with features expected from Release 17. The ongoing standardization in 3GPP regarding the integration of TSN and 5G systems is also addressed by Striffler et al. [78]. The authors identify open issues that still need to be addressed in terms of time synchronization, session continuity, and scheduling of different traffic streams.

On the other hand, there are several works focused on reviewing TSN and its potential use cases [11,153-155]. Lo Bello et al. [11] surveys TSN in industrial communication and automation systems while discussing core TSN standards and novel features which make TSN an enabler for several cutting-edge technologies. While Lo Bello et al. [11] remain generic, Samii et al. [154] focus on the automotive domain. They review the TSN standards in light of possible use cases in automotive systems. In addition, Deng et al. [155] also takes the automotive use case as an example to discuss the application of TSN in automobiles. The aim of the article is to provide an overview of recent advances and future trends in real-time Ethernet modeling and design methodologies for AVB and TSN. It surveys the current state of the field and provides references for researchers who are interested in this area. Moreover, Craciunas et al. [153] overviews the scheduling problem arising from time-sensitive network technologies like TTEthernet and TSN. The authors describe the main differences between two technologies, and they also describe the scheduling constraints that enable real-time temporal behavior on the level of individual communication streams. Reviewing the existing surveys on TSN, we observe that none of them is focusing on the converged TSN-5G system.

A comprehensive survey of the IEEE TSN and IETF DetNet standards targeting the support for ultra-low latency (ULL) is presented by Nasrallah et al. [7]. This work provides an in-deep survey of the development of IEEE TSN standards and highlights significant milestones illustrating the shift from Audio Video Bridging (AVB) to TSN. Flow synchronization, flow management, flow control, and flow integrity are some of the addressed aspects of TSN including:

- · Generic Precision Time Protocol to accomplish time synchronization of data
- · YANG data models to provide a framework for periodic status reporting and the configuration of bridges.
- · Resource Allocation Protocol (RAP) to achieve a distributed TSN Control Model since SRP is restricted to A/V applications having a limited number of SR classes. RAP improves scalability by leveraging the Link-Local Reservation Protocol (LRP) to support all TSN features.
- · Gate Control List used for the TSN flow control and IEEE 802.10bv Time-Aware Shaper (TAS) which is applicable for ULL requirements when all time-triggered windows are synchronized.

On the other hand, this study also addresses the key components in 5G standards for supporting ULL mechanisms. Some of the surveyed components are the Common Public Radio Interface (CPRI) which provides the specifications for packing and transporting baseband time domain and eCPRI to reduce the effective data rate. In addition, this survey also presents an overview of the main ULL research directions in the 5G wireless access segment and on TSN network. This survey covers the link and network layer latency reduction standards covering studies up to July 2018, while B. Briscoe et al. [156] surveys general techniques for reducing latencies in Internet Protocol (IP) packet networks covering studies up to August 2014.

Furthermore, there is also a systematic review of URLLC (Ultra-Reliable Latency Communication) technology of 3GPP presented in

⁴ https://github.com/zenepe1/TSN-5GReplicationPackage.

[157]. URLLC is a feature that will be covered by 5G and beyond 5G. This review analyzes the URLLC networking trend in wireless and wired communication. It mentions four technologies: Near area time deterministic wired network (IEEE TSN, 802.1Qx), Mobile time deterministic network (5G TSC, TSN-TT), broad area deterministic wired network (IETF DetNet) and metro deterministic wired network (OIF FlexE, ITU-T SG15 G.mtn). This study also surveys all the patents related to low-latency technology, patents related to high-reliability technology, and patents related to mobile network technology, showing the extreme increase in the number of patents based on 3GPP specifications.

Time Synchronization is an important part of latency in low-latency applications. All gPTP systems exchange timing information between different network devices on the control plane. The load created in the control plane due to the time synchronization can have a great impact on low-latency applications. A solution to mitigate the load created in the control plane is to use a centralized time synchronization system where timing information messages are exchanged only between a central controller and individual network devices. This approach is similar to software-defined networking (SDN) [158,159] even though SDN technology in wired and fixed network is more advanced than the SDN-based mobile network developments [160].

Another review of URLLC as a key enabler of mission-critical services is presented in [20]. It overviews the state of the art of URLLC in the physical layer, link layer, and network layer summarizing the potential implementation methods of URLLC. In addition, this study also illustrates the challenges of mobile systems to support the integration of URLLC technology and identifies the need for meaningful models to fit practical scenarios. All these studies have surveyed the 3GPP Releases including the key components of 5G standards [161] and TSN standards [11,162] separating them from each other.

5G-ACIA [18] aims at supporting 5G in the industrial domain and provides an insight overview of 5G in industrial applications, including the possible integration concepts and migration paths. There are a few white papers published by this organization that outline the critical requirements for interoperability and features of 5G. The integration of TSN and 5G is also addressed by this forum. The forum also explores how and why should the integration of TSN and 5G be applied in the industry. Their baseline is the 3GPP Release 16 for 5G specifications and IEEE standards for TSN specifications. We consider this paper since it provides not only an overview of the standards but also shows the TSN and 5G integration for various industrial automation use cases, i.e., controller-to-controller, controller-to-device, and device-to-compute communication.

Moreover, Parvez et al. [163] present several latency-critical applications which need to be supported by 5G. They also demonstrate the typical latency and data rate requirements for different mission-critical services, e.g., factory automation, robotics, virtual reality, and healthcare. Various solutions on RAN, Core Network, or caching solutions, are used to achieve low latency on a 5G system. 5G is also a promising solution for autonomous driving meeting the connectivity requirements of V2X communication for higher levels of autonomy [164].

The usage scenarios of 5G are also presented by Navarro-Ortiz et al. [165]. They present the most significant use cases expected for 5G including their scenarios and traffic models. Although this survey is focusing only on 5G, it performs useful analyses not only on the characteristics and requirements for 5G communications but also on 5G usage scenarios allowing 5G stakeholders and researchers to evaluate the performance of 5G solutions under the most critical requirements. Furthermore, 5G should be adapted to a wide range of scenarios such as indoor, urban, suburban, rural areas, etc, which will set new requirements for 5G channel modeling [166]. In addition, Ai et al. [161] have identified significant 5G-based key technologies for high-speed railways to develop innovative communication network architectures that ensure high-quality transmissions for both passengers and railway operations and control systems. Recse et al. [167] and Kaloxylos [168] advocate network slicing to be the key enabler to realize 5G in IoT. The use of the network slicing method can effectively guarantee the QoS requirements of different services by splitting the existing physical network to form multiple independent logical networks with customized services [169]. Even though the integration of TSN and 5G is not considered by the authors, the investigated technologies can provide a good foundation for converged wired and wireless architecture considering 5G. However, the research on the application of network slicing over TSN networks [82] is still in its infancy and the telecommunication organizations are still working on the standardization of such technology.

Scanzio, Wisniewski, and Gaj [170] perform an analysis of the state of the art in the area of heterogeneous industrial networks. This survey investigates both wired and wireless technologies considering technological aspects and performance targets, e.g., dependability. It also highlights the main challenges and communication requirements of industrial applications. 5G is also one of its targeted wireless technologies but the integration of wired and wireless technologies is still a challenge that they aim to consider as their future work.

Although 3GPP standards offer the possibility to converge TSN and 5G networks, a comprehensive overview of TSN-5G integration scenarios and a structured research map of the area is still missing. In this context, with this systematic literature review, we attempt to present all the current studies conducted in the scope of TSN-5G integration, while identifying the gaps in the existing research and highlighting further research opportunities for researchers and practitioners.

9. Conclusions

In this article, we presented the planning, execution, and results of a Systematic Literature Review (SLR) on the integration of TSN and 5G technologies. The SLR provides a holistic overview and structured map of the state of the art research on TSN-5G integration. We identified 189 research studies in the initial phase of search and selection. After several refinements, we selected 82 of them as the primary studies that focus on the integration of TSN and 5G technologies. We extracted the required data from these primary studies by using a well-defined and thorough data extraction process. The extracted data was then analyzed and synthesized to answer the three research questions posed in this SLR.

The first research question was answered by analyzing the primary studies according to their technical contributions. We noticed that 74% of the studies follow the architecture proposed by the 3GPP working group, while still encountering difficulties on time synchronization, resource management, and flow management of the integrated system. Furthermore, the most commonly used time synchronization approach is the transparent clock approach that is proposed in the 3GPP specification. In addition, 72% (59 studies) of all the primary studies provide software design, while only 12 primary studies (15% of the total) present hardware design for the TSN and 5G integration. The majority of the primary studies (29 studies) use simulated networks for their empirical evaluation, while 19 studies use a real network to perform the evaluation of their TSN-5G suggested technique/approach.

The second research question was answered by classifying the primary studies according to well-defined classification criteria and showing the research trends in the area of TSN-5G integration. The results show that the interest of researchers in the area is continuously growing. The IEEE International Workshop on Factory Communication Systems (WFCS) and the IEEE International Conference on Emerging Technologies and Factory Automation (ETFA) have published the highest number of primary studies to date.

To answer the third research question, we used horizontal analysis to investigate the relationship between various sets of categories in the proposed classification. This analysis resulted in the identification of potential gaps in the research area and opportunities for future research. Among the others, it calls for the research community to develop prototypes of a tool that implements the scientific techniques for the integration of TSN and 5G technologies.

The results of this study are comprehensive enough for researchers and practitioners in identifying current research trends, potential gaps, and future research directions in the context of integrating TSN and 5G technologies. In the future, we plan to provide an approach for timing synchronization between TSN and 5G following the integration architecture suggested by the 3GPP working group. Another contribution to the research community would be to evaluate the timing approach using one of the well-known simulation frameworks, named OMNet++.

Declaration of competing interest

The authors declare the following financial interests/personal relationships which may be considered as potential competing interests: Zenepe Satka reports financial support was provided by Sweden's Innovation Agency.

Zenepe Satka reports a relationship with Sweden's Innovation Agency that includes: funding grants.

Data availability

No data was used for the research described in the article.

Acknowledgments

The work in this paper is supported by the Swedish Governmental Agency for Innovation Systems (VINNOVA) via the PROVIDENT, DESTINE, and INTERCONNECT projects, and the Swedish Knowledge Foundation via the DPAC and HERO projects. We would like to thank all our industrial partners, especially Arcticus Systems, HIAB, and Volvo Construction Equipment.

References

- M. Wollschlaeger, T. Sauter, J. Jasperneite, The future of industrial communication: Automation networks in the era of the internet of things and industry 4.0, IEEE Ind. Electron. Mag. 11 (1) (2017) 17–27, http://dx.doi.org/10.1109/ MIE.2017.2649104.
- [2] A. Aijaz, M. Sooriyabandara, The tactile internet for industries: A review, Proc. IEEE 107 (2) (2019) 414–435, http://dx.doi.org/10.1109/JPROC.2018. 2878265.
- [3] C.S.V. Gutiérrez, L.U.S. Juan, I.Z. Ugarte, V.M. Vilches, Time-sensitive networking for robotics, 2018, arXiv preprint arXiv:1804.07643.
- [4] A. Kostrzewa, R. Ernst, Achieving safety and performance with reconfiguration protocol for ethernet TSN in automotive systems, J. Syst. Archit. 118 (2021) 102208, http://dx.doi.org/10.1016/j.sysarc.2021.102208.
- [5] F. Prinz, M. Schoeffler, A. Lechler, A. Verl, Dynamic real-time orchestration of I4.0 components based on time-sensitive networking, Procedia CIRP 72 (2018) 910–915.
- [6] M. Ashjaei, L.L. Bello, M. Daneshtalab, G. Patti, S. Saponara, S. Mubeen, Time-Sensitive Networking in automotive embedded systems: State of the art and research opportunities, J. Syst. Archit. 117 (2021) 102137, http://dx.doi.org/ 10.1016/j.sysarc.2021.102137.
- [7] A. Nasrallah, A.S. Thyagaturu, Z. Alharbi, C. Wang, X. Shao, M. Reisslein, H. ElBakoury, Ultra-low latency (ULL) networks: The IEEE TSN and IETF DetNet standards and related 5G ULL research, IEEE Commun. Surv. Tutor. 21 (1) (2019) 88–145, http://dx.doi.org/10.1109/COMST.2018.2869350.
- [8] L. Deng, X. Xiao, H. Liu, R. Li, G. Xie, A low-delay AVB flow scheduling method occupying the guard band in Time-Sensitive Networking, J. Syst. Archit. 129 (2022) 102586, http://dx.doi.org/10.1016/j.sysarc.2022.102586.
- [9] M. Ashjaei, M. Sjödin, S. Mubeen, A novel frame preemption model in TSN networks, J. Syst. Archit. 116 (2021) 102037, http://dx.doi.org/10.1016/ j.sysarc.2021.102037, URL: https://www.sciencedirect.com/science/article/pii/ \$1383762121000382.
- [10] S. Mubeen, L. Lo Bello, M. Daneshtalab, S. Saponara, Guest Editorial: Special issue on parallel, distributed, and network-based processing in next-generation embedded systems, J. Syst. Archit. 117 (2021) 102159, http://dx.doi.org/ 10.1016/j.sysarc.2021.102159, URL: https://www.sciencedirect.com/science/ article/pii/S1383762121001144.

- [11] L. Lo Bello, W. Steiner, A perspective on IEEE time-sensitive networking for industrial communication and automation systems, Proc. IEEE 107 (6) (2019) 1094–1120, http://dx.doi.org/10.1109/JPROC.2019.2905334.
- [12] J.G. Andrews, S. Buzzi, W. Choi, S.V. Hanly, A. Lozano, A.C.K. Soong, J.C. Zhang, What will 5G be? IEEE J. Sel. Areas Commun. 32 (6) (2014) 1065–1082, http://dx.doi.org/10.1109/JSAC.2014.2328098.
- [13] T. Yang, S. Wang, B. Zhan, N. Zhan, J. Li, S. Xiang, Z. Xiang, B. Mao, Formal analysis of 5G authentication and key management for applications (AKMA), J. Syst. Archit. 126 (2022) 102478, http://dx.doi.org/10.1016/j.sysarc.2022. 102478.
- [14] 3GPP TS 23.501, System Architecture for the 5G System, 2020, V. 16.5.1.
- [15] 3GPP TS 33.501, Security Architecture and Procedures for 5G System, 2020, V. 16.3.0.
- [16] 3GPP TS 38.331, Radio Resource Control (RRC) Protocol Specification, 2020, V. 16.1.0.
- [17] 3GPP TR 22.832, Study on Enhancements for Cyber-Physical Control Applications in Vertical Domains, 2019, V. 17.2.0.
- [18] 5G ACIA, 5G for Connected Industries and Automation, white paper, 2019, URL: https://5g-acia.org/resources/whitepapers-deliverables/.
- [19] M. Bennis, M. Debbah, H.V. Poor, Ultrareliable and low-latency wireless communication: Tail, risk, and scale, Proc. IEEE 106 (10) (2018) 1834–1853, http://dx.doi.org/10.1109/JPROC.2018.2867029.
- [20] D. Feng, L. Lai, J. Luo, Y. Zhong, C. Zheng, K. Ying, Ultra-reliable and lowlatency communications: applications, opportunities and challenges, Sci. China Inf. Sci. 64 (2021) 1–12.
- [21] B. Kitchenham, P. Brereton, A systematic review of systematic review process research in software engineering, Inf. Softw. Technol. 55 (12) (2013) 2049–2075, http://dx.doi.org/10.1016/j.infsof.2013.07.010.
- [22] F. Ciccozzi, L. Addazi, S.A. Asadollah, B. Lisper, A.N. Masud, S. Mubeen, A comprehensive exploration of languages for parallel computing, ACM Comput. Surv. 55 (2) (2022) http://dx.doi.org/10.1145/3485008.
- [23] M. Khoshnevisan, V. Joseph, P. Gupta, F. Meshkati, R. Prakash, P. Tinnakornsrisuphap, 5G industrial networks with CoMP for URLLC and time sensitive network architecture, IEEE J. Sel. Areas Commun. 37 (4) (2019) 947–959, http://dx.doi.org/10.1109/JSAC.2019.2898744.
- [24] D. Cavalcanti, J. Perez-Ramirez, M.M. Rashid, J. Fang, M. Galeev, K.B. Stanton, Extending accurate time distribution and timeliness capabilities over the air to enable future wireless industrial automation systems, Proc. IEEE 107 (6) (2019) 1132–1152, http://dx.doi.org/10.1109/JPROC.2019.2903414.
- [25] I. Godor, M. Luvisotto, S. Ruffini, K. Wang, D. Patel, J. Sachs, O. Dobrijevic, D.P. Venmani, O.L. Moult, J. Costa-Requena, A. Poutanen, C. Marshall, J. Farkas, A look inside 5G standards to support time synchronization for smart manufacturing, IEEE Commun. Stand. Mag. 4 (3) (2020) 14–21, http://dx.doi. org/10.1109/MCOMSTD.001.2000010.
- [26] K. Petersen, R. Feldt, S. Mujtaba, M. Mattsson, Systematic mapping studies in software engineering, in: Proceedings of the 12th International Conference on Evaluation and Assessment in Software Engineering, EASE '08, BCS Learning & Development Ltd., Swindon, GBR, 2008, pp. 68–77.
- [27] P. Brereton, B.A. Kitchenham, D. Budgen, M. Turner, M. Khalil, Lessons from applying the systematic literature review process within the software engineering domain, J. Syst. Softw. 80 (4) (2007) 571–583.
- [28] S. Fabbri, C. Silva, E. Hernandes, F. Octaviano, A. Di Thommazo, A. Belgamo, Improvements in the StArt tool to better support the systematic review process, in: Proceedings of the 20th International Conference on Evaluation and Assessment in Software Engineering, EASE '16, Association for Computing Machinery, New York, NY, USA, 2016, http://dx.doi.org/10.1145/2915970.2916013.
- [29] V. Garousi, M. Felderer, M.V. Mäntylä, Guidelines for including grey literature and conducting multivocal literature reviews in software engineering, Inf. Softw. Technol. 106 (2019) 101–121.
- [30] C. Wohlin, Guidelines for snowballing in systematic literature studies and a replication in software engineering, in: Proceedings of the 18th International Conference on Evaluation and Assessment in Software Engineering, EASE '14, Association for Computing Machinery, New York, NY, USA, 2014, http://dx.doi. org/10.1145/2601248.2601268, URL: https://doi-org.ep.bib.mdh.se/10.1145/ 2601248.2601268.
- [31] K. Charmaz, L.L. Belgrave, Grounded theory, in: The Blackwell Encyclopedia of Sociology, 2007.
- [32] M.K. Atiq, R. Muzaffar, Ó. Seijo, I. Val, H.-P. Bernhard, When IEEE 802.11 and 5G meet time-sensitive networking, IEEE Open J. Ind. Electron. Soc. 3 (2022) 14–36, http://dx.doi.org/10.1109/OJIES.2021.3135524.
- [33] IEEE Standard 802.1Qcc-2018, IEEE Standard for Local and Metropolitan Area Networks–Bridges and Bridged Networks–Amendment 31: Stream Reservation Protocol (SRP) Enhancements and Performance Improvements, 2018.
- [34] M. Bjorklund, The YANG 1.1 Data Modeling Language, RFC 7950, 2016.

- [35] R. Enns, M. Bjorklund, J. Schoenwaelder, A. Bierman, Network Configuration Protocol (NETCONF), RFC 6241, 2016.
- [36] I. Álvarez, A. Servera, J. Proenza, M. Ashjaei, S. Mubeen, Implementing a first CNC for scheduling and configuring TSN networks, in: 2022 IEEE 27th International Conference on Emerging Technologies and Factory Automation, ETFA, 2022, pp. 1–4, http://dx.doi.org/10.1109/ETFA52439.2022.9921518.
- [37] A. Bierman, M. Bjorklund, K. Watsen, RESTCONF Protocol, RFC 8040, 2017.
- [38] IEEE Standard for Local and Metropolitan Area Networks Virtual Bridged Local Area Networks - Amendment 07: Multiple Registration Protocol, Std 802.1ak-2007 (Amendment to IEEE Std 802.1QTM-2005), 2007, pp. 1–115, http://dx.doi.org/10.1109/IEEESTD.2007.380667.
- [39] 3GPP, Study on 5GS Enhanced support of Vertical and LAN Services (Release 16), 3rd Generation Partnership Project (3GPP), 2019, TR 23.734, v16.1.0..
- [40] IEEE Std 1588-2008, IEEE Standard for a Precision Clock Synchronization Protocol for Networked Measurement and Control Systems, 2008.
- [41] N. Kerö, A. Puhm, T. Kernen, A. Mroczkowski, Performance and reliability aspects of clock synchronization techniques for industrial automation, Proc. IEEE 107 (6) (2019) 1011–1026, http://dx.doi.org/10.1109/JPROC.2019.2915972.
- [42] O. Michel, R. Bifulco, G. Rétvári, S. Schmid, The programmable data plane: Abstractions, architectures, algorithms, and applications, ACM Comput. Surv. 54 (4) (2021) http://dx.doi.org/10.1145/3447868.
- [43] N. Foster, N. McKeown, J. Rexford, G. Parulkar, L. Peterson, O. Sunay, Using deep programmability to put network owners in control, ACM SIGCOMM Comput. Commun. Rev. 50 (4) (2020) 82–88, http://dx.doi.org/10.1145/3431832. 3431842, URL: https://doi-org.ep.bib.mdh.se/10.1145/3431832.3431842.
- [44] J.P. Thomesse, Fieldbuses and interoperability, Control Eng. Pract. 7 (1) (1999) 81–94.
- [45] R. Wang, C. Gu, S. He, Z. Shi, W. Meng, An interoperable and flat Industrial Internet of Things architecture for low latency data collection in manufacturing systems, J. Syst. Archit. 129 (2022) 102631, http://dx.doi.org/10.1016/j.sysarc. 2022.102631.
- [46] Y. Katsube, K. Nagami, S. Matsuzawa, H. Esaki, Internetworking based on cell switch router-architecture and protocol overview, Proc. IEEE 85 (12) (1997) 1998–2006, http://dx.doi.org/10.1109/5.650181.
- [47] Z. Bakhshi, G. Rodriguez-Navas, A preliminary roadmap for dependability research in fog computing, ACM SIGBED Rev. 16 (4) (2020) 14–19, http: //dx.doi.org/10.1145/3378408.3378410.
- [48] M. Manderscheid, G. Weiss, R. Knorr, Verification of network end-to-end latencies for adaptive ethernet-based cyber-physical systems, J. Syst. Archit. 88 (2018) 23–32, http://dx.doi.org/10.1016/j.sysarc.2018.05.004.
- [49] S. Mubeen, E. Lisova, A. Vulgarakis Feljan, Timing predictability and security in safety-critical industrial cyber-physical systems: A position paper, Appl. Sci. 10 (9) (2020) 3125.
- [50] Y. He, D. She, S. Stuijk, H. Corporaal, Efficient communication support in predictable heterogeneous MPSoC designs for streaming applications, J. Syst. Archit. 59 (10, Part A) (2013) 878–888, http://dx.doi.org/10.1016/j.sysarc. 2013.04.005, Smart Camera Architecture.
- [51] W. Wang, H. Huang, L. Xue, Q. Li, R. Malekian, Y. Zhang, Blockchainassisted handover authentication for intelligent telehealth in multi-server edge computing environment, J. Syst. Archit. 115 (2021) 102024, http://dx.doi.org/ 10.1016/j.sysarc.2021.102024.
- [52] Ericsson Blog, This is the key to mobility robustness in 5G networks, 2020, [Online]. Available: https://www.ericsson.com/en/blog/2020/5/the-keyto-mobility-robustness-5g-networks.
- [53] M. Louvel, A. Plantec, J.-P. Babau, Resource management for multimedia applications, distributed in open and heterogeneous home networks, J. Syst. Archit. 59 (3) (2013) 121–134, http://dx.doi.org/10.1016/j.sysarc.2013.01.003.
- [54] Z. Sanaei, S. Abolfazli, A. Gani, R. Buyya, Heterogeneity in mobile cloud computing: Taxonomy and open challenges, IEEE Commun. Surv. Tutor. 16 (1) (2014) 369–392.
- [55] V.-P. Ranganath, P. Vallathol, P. Gupta, Compatibility testing using patternsbased trace comparison, in: Proceedings of the 29th ACM/IEEE International Conference on Automated Software Engineering, ASE '14, Association for Computing Machinery, New York, NY, USA, 2014, pp. 469–478, http://dx.doi. org/10.1145/2642937.2642942.
- [56] M.A. Kafi, J.B. Othman, N. Badache, A survey on reliability protocols in wireless sensor networks, ACM Comput. Surv. 50 (2) (2017) http://dx.doi.org/10.1145/ 3064004.
- [57] Z. Feng, Q. Deng, M. Cai, J. Li, Efficient reservation-based fault-tolerant scheduling for IEEE 802.1Qbv time-sensitive networking, J. Syst. Archit. 123 (2022) 102381, http://dx.doi.org/10.1016/j.sysarc.2021.102381.
- [58] S.T. Redwine, W.E. Riddle, Software technology maturation, in: Proceedings of the 8th International Conference on Software Engineering, ICSE '85, IEEE Computer Society Press, Washington, DC, USA, 1985, pp. 189–200.

- [59] E. Kyriakakis, M. Lund, L. Pezzarossa, J. Sparsø, M. Schoeberl, A timepredictable open-source TTEthernet end-system, J. Syst. Archit. 108 (2020) 101744, http://dx.doi.org/10.1016/j.sysarc.2020.101744, URL: https://www. sciencedirect.com/science/article/pii/S1383762120300382.
- [60] A. Larrañaga, M.C. Lucas-Estañ, I. Martinez, I. Val, J. Gozalvez, Analysis of 5G-TSN integration to support industry 4.0, in: 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation, ETFA, vol. 1, IEEE, Vienna, Austria, 2020, pp. 1111–1114.
- [61] D. Ginthör, J. von Hoyningen-Huene, R. Guillaume, H. Schotten, Analysis of multi-user scheduling in a TSN-enabled 5G system for industrial applications, in: 2019 IEEE International Conference on Industrial Internet, ICII, IEEE, Orlando, FL, USA, 2019, pp. 190–199, http://dx.doi.org/10.1109/ICII.2019.00044.
- [62] D. Wang, T. Sun, Leveraging 5G TSN in V2X communication for cloud vehicle, in: 2020 IEEE International Conference on Edge Computing, EDGE, IEEE, Beijing, China, 2020, pp. 106–110.
- [63] R. Lu, L. Zhang, J. Ni, Y. Fang, 5G Vehicle-to-Everything Services: Gearing Up for Security and Privacy, Proc. IEEE 108 (2) (2020) 373–389, http://dx.doi. org/10.1109/JPROC.2019.2948302.
- [64] Y. Ni, L. Cai, J. He, A. Vinel, Y. Li, H. Mosavat-Jahromi, J. Pan, Toward reliable and scalable internet of vehicles: Performance analysis and resource management, Proc. IEEE 108 (2) (2020) 324–340, http://dx.doi.org/10.1109/ JPROC.2019.2950349.
- [65] R. Wieringa, N. Maiden, N. Mead, C. Rolland, Requirements engineering paper classification and evaluation criteria: a proposal and a discussion, Requir. Eng. 11 (1) (2006) 102–107.
- [66] S. Mubeen, S.A. Asadollah, A.V. Papadopoulos, M. Ashjaei, H. Pei-Breivold, M. Behnam, Management of service level agreements for cloud services in IoT: A systematic mapping study, IEEE Access 6 (2018) 30184–30207, http: //dx.doi.org/10.1109/ACCESS.2017.2744677.
- [67] D.S. Cruzes, T. Dyba, Recommended steps for thematic synthesis in software engineering, in: 2011 International Symposium on Empirical Software Engineering and Measurement, IEEE, Banff, AB, Canada, 2011, pp. 275–284, http: //dx.doi.org/10.1109/ESEM.2011.36.
- [68] D. King, A. Farrel, A PCE-based framework for future internet deterministic and time-sensitive networks, in: 2020 16th International Conference on Network and Service Management, CNSM, IEEE, Izmir, Turkey, 2020, pp. 1–7, http: //dx.doi.org/10.23919/CNSM50824.2020.9269125.
- [69] J. Zou, S. Adrian Sasu, M. Lawin, A. Dochhan, J.-P. Elbers, M. Eiselt, Advanced optical access technologies for next-generation (5G) mobile networks [Invited], J. Opt. Commun. Netw. 12 (10) (2020) D86–D98, http://dx.doi.org/10.1364/ JOCN.391033.
- [70] S. Math, L. Zhang, S. Kim, I. Ryoo, An intelligent real-time traffic control based on mobile edge computing for individual private environment, Secur. Commun. Netw. 2020 (2020) 8881640, URL: https://doi.org/10.1155/2020/8881640.
- [71] M. Yang, S. Lim, S.-M. Oh, J. Shin, An uplink transmission scheme for TSN service in 5G industrial IoT, in: 2020 International Conference on Information and Communication Technology Convergence, ICTC, IEEE, Jeju, Korea (South), 2020, pp. 902–904, http://dx.doi.org/10.1109/ICTC49870.2020.9289303.
- [72] J. Ohms, M. Böhm, D. Wermser, Concept of a TSN to real-time wireless gateway in the context of 5G URLLC, in: 2020 8th International Conference on Wireless Networks and Mobile Communications, WINCOM, IEEE, Reims, France, 2020, pp. 1–6, http://dx.doi.org/10.1109/WINCOM50532.2020.9272460.
- [73] Y. Liu, Y. Zhou, J. Yuan, L. Liu, Delay aware flow scheduling for time sensitive fronthaul networks in centralized radio access network, IEEE Trans. Commun. 68 (5) (2020) 2992–3009, http://dx.doi.org/10.1109/TCOMM.2020.2976062.
- [74] D. Ginthör, R. Guillaume, J. von Hoyningen-Huene, M. Schüngel, H.D. Schotten, End-to-end optimized joint scheduling of converged wireless and wired timesensitive networks, in: 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation, ETFA, vol. 1, IEEE, Vienna, Austria, 2020, pp. 222–229.
- [75] J. Zou, S.A. Sasu, J. Messenger, J.-P. Elbers, Options for time-sensitive networking for 5G fronthaul, in: 45th European Conference on Optical Communication (ECOC 2019), IET, Dublin, Ireland, 2019, pp. 1–3.
- [76] C. Mannweiler, B. Gajic, P. Rost, R.S. Ganesan, C. Markwart, R. Halfmann, J. Gebert, A. Wich, Reliable and deterministic mobile communications for industry 4.0: Key challenges and solutions for the integration of the 3GPP 5G system with IEEE, in: Mobile Communication Technologies and Applications; 24. ITG-Symposium, VDE, Osnabrueck, Germany, 2019, pp. 1–6.
- [77] R.B. Abreu, G. Pocovi, T.H. Jacobsen, M. Centenaro, K.I. Pedersen, T.E. Kolding, Scheduling enhancements and performance evaluation of downlink 5G timesensitive communications, IEEE Access 8 (2020) 128106–128115, http://dx. doi.org/10.1109/ACCESS.2020.3008598.
- [78] T. Striffler, N. Michailow, M. Bahr, Time-sensitive networking in 5th generation cellular networks - Current state and open topics, in: 2nd IEEE 5G World Forum (5GWF), IEEE, Dresden, Germany, 2019, pp. 547–552.

- [79] A. Neumann, L. Wisniewski, R.S. Ganesan, P. Rost, J. Jasperneite, Towards integration of Industrial Ethernet with 5G mobile networks, in: 2018 14th IEEE International Workshop on Factory Communication Systems, WFCS, IEEE, Imperia, Italy, 2018, pp. 1–4, http://dx.doi.org/10.1109/WFCS.2018.8402373.
- [80] E. Genc, L.F. Del Carpio, Wi-Fi QoS enhancements for downlink operations in industrial automation using TSN, in: 2019 15th IEEE International Workshop on Factory Communication Systems, WFCS, IEEE, Sundsvall, Sweden, 2019, pp. 1–6, http://dx.doi.org/10.1109/WFCS.2019.8757992.
- [81] J. Prados-Garzon, T. Taleb, Asynchronous time-sensitive networking for 5G backhauling, IEEE Netw. 35 (2) (2021) 144–151, http://dx.doi.org/10.1109/ MNET.011.2000402.
- [82] S. Bhattacharjee, K. Katsalis, O. Arouk, R. Schmidt, T. Wang, X. An, T. Bauschert, N. Nikaein, Network slicing for TSN-based transport networks, IEEE Access 9 (2021) 62788–62809, http://dx.doi.org/10.1109/ACCESS.2021. 3074802.
- [83] J. Farkas, B. Varga, G. Miklós, J. Sachs, 5G-TSN Integration Meets Networking Requirements for Industrial Automation, Ericsson Technol. Rev. 96 (7) (2019) 45–51.
- [84] F. Song, L. Li, I. You, H. Zhang, Enabling heterogeneous deterministic networks with smart collaborative theory, IEEE Netw. 35 (3) (2021) 64–71, http://dx. doi.org/10.1109/MNET.011.2000613.
- [85] J. Prados-Garzon, T. Taleb, M. Bagaa, Optimization of flow allocation in asynchronous deterministic 5G transport networks by leveraging data analytics, IEEE Trans. Mob. Comput. (2021) http://dx.doi.org/10.1109/TMC.2021.3099979.
- [86] D. Ginthör, R. Guillaume, M. Schüngel, H.D. Schotten, Robust end-to-end schedules for wireless time-sensitive networks under correlated large-scale fading, in: 2021 17th IEEE International Conference on Factory Communication Systems, WFCS, IEEE, Linz, Austria, 2021, pp. 115–122, http://dx.doi.org/10. 1109/WFCS46889.2021.9483589.
- [87] N. Shibata, P. Zhu, K. Nishimura, Y. Yoshida, K. Hayashi, M. Hirota, R. Harada, K. Honda, S. Kaneko, J. Terada, K.-I. Kitayama, Time sensitive networking for 5G NR fronthauls and massive IoT traffic, J. Lightwave Technol. 39 (16) (2021) 5336–5343.
- [88] D. Cavalcanti, Wireless TSN Definitions, Use Cases & Standards Roadmap, Avnu Alliance White Paper, 2020, URL: https://avnu.org/wireless-tsn-paper/.
- [89] 5G-ACIA, Integration of 5G with Time-Sensitive Networking for Industrial Communications, white paper, 2021, URL: https://5gacia.org/whitepapers/integration-of-5g-with-time-sensitive-networking-forindustrial-communications/.
- [90] G. Brown, Ultra-Reliable Low-Latency 5G for Industrial Automation, Qualcomm White Paper, 2019, URL: https://www.qualcomm.com/media/documents/files/ read-the-white-paper-by-heavy-reading.pdf.
- [91] R. Petre, Testing time-sensitive networking over 5G: Time synchronization (802.1as), 2021, URL: https://www.spirent.com/blogs/testing-time-sensitivenetworking-over-5g-time-synchronization, Accessed: 2021-11-20.
- [92] Y. Zhang, Q. Xu, M. Li, C. Chen, X. Guan, QoS-aware mapping and scheduling for virtual network functions in industrial 5G-TSN network, in: 2021 IEEE Global Communications Conference, GLOBECOM, IEEE, Madrid, Spain, 2021, pp. 1–6, http://dx.doi.org/10.1109/GLOBECOM46510.2021.9685749.
- [93] H. Xu, J. Xin, S. Xu, H. Zhang, RAN enhancement to support propagation delay compensation of TSN, in: 2021 IEEE 9th International Conference on Information, Communication and Networks, ICICN, IEEE, Xi'an, China, 2021, pp. 179–182, http://dx.doi.org/10.1109/ICICN52636.2021.9674019.
- [94] D. Patel, J. Diachina, S. Ruffini, M. De Andrade, J. Sachs, D.P. Venmani, Time error analysis of 5G time synchronization solutions for time aware industrial networks, in: 2021 IEEE International Symposium on Precision Clock Synchronization for Measurement, Control, and Communication, ISPCS, IEEE, NA, FL, USA, 2021, pp. 1–6, http://dx.doi.org/10.1109/ISPCS49990.2021.9615318.
- [95] O. Akudo Nwogu, G. Diaz, M. Abdennebi, Differential traffic QoS scheduling for 5G/6G fronthaul networks, in: 2021 31st International Telecommunication Networks and Applications Conference, ITNAC, IEEE, Sydney, 2021, pp. 113–120, http://dx.doi.org/10.1109/ITNAC53136.2021.9652162.
- [96] Z. Liu, H. Wang, H. Sun, J. Su, B. Zhang, M. Liu, R. Zhu, Software defined network based 5G and time- sensitive network fusion for power services with ultra-low latency requirements, in: 2021 International Conference on Wireless Communications and Smart Grid, ICWCSG, IEEE, Hangzhou, China, 2021, pp. 381–385, http://dx.doi.org/10.1109/ICWCSG53609.2021.00082.
- [97] Z. Chai, W. Liu, M. Li, J. Lei, Cross domain clock synchronization based on data packet relay in 5G-TSN integrated network, in: 2021 IEEE 4th International Conference on Electronics and Communication Engineering, ICECE, IEEE, Xi'an, China, 2021, pp. 141–145, http://dx.doi.org/10.1109/ICECE54449.2021. 9674640.
- [98] C.-R. Lorena, J. Prados-Garzon, P. Ameigeiras, P. Muñoz, J.M. Lopez-Soler, 5G infrastructure network slicing: E2E mean delay model and effectiveness assessment to reduce downtimes in industry 4.0, Sensors 22 (1) (2022) 229, URL: https://doi.org/10.3390/s22010229.

- [99] Y. Zhang, Q. Xu, C. Chen, M. Li, Wireless/wired integrated transmission for industrial cyber-physical systems: risk-sensitive co-design of 5G and TSN protocols, Sci. China Inf. Sci. 65 (2022) URL: https://doi.org/10.1007/s11432-020-3344-8.
- [100] W. Lei, A.C.K. Soong, L. Jianghua, W. Yong, B. Classon, W. Xiao, D. Mazzarese, Z. Yang, T. Saboorian, 5G industrial IoT, in: Wireless Networks, 2021, pp. 515–532, http://dx.doi.org/10.1007/978-3-030-73703-09.
- [101] Z. Satka, D. Pantzar, A. Magnusson, M. Ashjaei, H. Fotouhi, M. Sjödin, M. Daneshtalab, S. Mubeen, Developing a translation technique for converged TSN-5G communication, in: 2022 IEEE 18th International Conference on Factory Communication Systems, WFCS, IEEE, 2022, pp. 1–8, http://dx.doi.org/10. 1109/WFCS53837.2022.9779191.
- [102] P. Kielhs, J. Ansari, M.H. Jafari, P. Becker, J. Sachs, N. König, A. Göppert, R.H. Schmitt, Prototype of 5G integrated with TSN for edge-controlled mobile robotics, Int. J. Electron. Telecommun. 11 (11) (2022) http://dx.doi.org/10. 3390/electronics11111666.
- [103] F. Luque-Schempp, L. Panizo, M.-d.-M. Gallardo, P. Merino, J. Rivas, Toward zero touch configuration of 5G non-public networks for time sensitive networking, IEEE Netw. 36 (2) (2022) 50–56, http://dx.doi.org/10.1109/MNET.006. 2100442.
- [104] K. Nikhileswar, K. Prabhu, D. Cavalcanti, Traffic steering in edge compute devices using express data path for 5G and TSN integration, in: 2022 IEEE 18th International Conference on Factory Communication Systems, WFCS, IEEE, 2022, pp. 1–6, http://dx.doi.org/10.1109/WFCS53837.2022.9779167.
- [105] J. Shepard, When will 5G enter industrial TSN?, 2022, URL: https://eu.mouser. com/applications/when-5g-enter-industrial-tsn/, Accessed: 2022-07-13.
- [106] P.M. Rost, T. Kolding, Performance of integrated 3GPP 5G and IEEE TSN networks, IEEE Commun. Stand. Mag. 6 (2) (2022) 51–56, http://dx.doi.org/ 10.1109/MCOMSTD.0001.2000013.
- [107] X. Li, Y. Chi, X. Jia, Y. Xie, Research on integration of 5G and TSN, in: Advances in Transdisciplinary Engineering, IOS Press, 2022, pp. 128–135.
- [108] G. Peng, S. Wang, Y. Huang, T. Huang, Y. Liu, Enabling deterministic tasks with multi-access edge computing in 5G networks, IEEE Commun. Mag. 60 (8) (2022) 36–42, http://dx.doi.org/10.1109/MCOM.001.2101073.
- [109] J. Yang, G. Yu, Traffic scheduling for 5G-TSN integrated systems, in: 2022 International Symposium on Wireless Communication Systems, ISWCS, 2022, pp. 1–6, http://dx.doi.org/10.1109/ISWCS56560.2022.9940254.
- [110] Y. Chen, X. Yao, Z. Gan, Z. You, P. Wu, W. Wang, Power service mapping scheduling method based on fusion of 5G and time-sensitive network, in: 2022 IEEE 6th Advanced Information Technology, Electronic and Automation Control Conference (IAEAC), 2022, pp. 1309–1314, http://dx.doi.org/10.1109/ IAEAC54830.2022.9929661.
- [111] K. Nikhileswar, K. Prabhu, D. Cavalcanti, A. Regev, Time-sensitive networking over 5G for industrial control systems, in: 2022 IEEE 27th International Conference on Emerging Technologies and Factory Automation, ETFA, 2022, pp. 1–8, http://dx.doi.org/10.1109/ETFA52439.2022.9921680.
- [112] H. Zhang, N. Lu, W. Xie, P. Li, Analysis of the use cases and technical enhancements of TSN, in: 2021 2nd Information Communication Technologies Conference, ICTC, IEEE, Nanjing, China, 2021, pp. 288–291, http://dx.doi.org/ 10.1109/ICTC51749.2021.9441589.
- [113] L. Martenvormfelde, A. Neumann, L. Wisniewski, J. Jasperneite, A simulation model for integrating 5G into time sensitive networking as a transparent bridge, in: 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation, ETFA, vol. 1, IEEE, Vienna, Austria, 2020, pp. 1103–1106.
- [114] Z. Satka, M. Ashjaei, H. Fotouhi, M. Daneshtalab, M. Sjödin, S. Mubeen, QoS-MAN: A novel QoS mapping algorithm for TSN-5G flows, in: 2022 IEEE 28th International Conference on Embedded and Real-Time Computing Systems and Applications, RTCSA, 2022, pp. 220–227, http://dx.doi.org/10. 1109/RTCSA55878.2022.00030.
- [115] M. Schüngel, S. Dietrich, D. Ginthör, S.-P. Chen, M. Kuhn, Analysis of time synchronization for converged wired and wireless networks, in: 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation, ETFA, vol. 1, IEEE, Vienna, Austria, 2020, pp. 198–205.
- [116] M. Gundall, C. Huber, P. Rost, R. Halfmann, H.D. Schotten, Integration of 5G with TSN as prerequisite for a highly flexible future industrial automation: Time synchronization based on IEEE 802.1AS, in: IECON 2020 the 46th Annual Conference of the IEEE Industrial Electronics Society, IEEE, Singapore, 2020, pp. 3823–3830, http://dx.doi.org/10.1109/IECON43393.2020.9254296.
- [117] M. Schüngel, S. Dietrich, D. Ginthör, S.-P. Chen, M. Kuhn, Single message distribution of timing information for time synchronization in converged wired and wireless networks, in: 2020 25th IEEE International Conference on Emerging Technologies and Factory Automation, ETFA, vol. 1, IEEE, Vienna, Austria, 2020, pp. 286–293.

- [118] S. Bhattacharjee, R. Schmidt, K. Katsalis, C.-Y. Chang, T. Bauschert, N. Nikaein, Time-sensitive networking for 5G fronthaul networks, in: ICC 2020 - 2020 IEEE International Conference on Communications, ICC, IEEE, Dublin, 2020, pp. 1–7, http://dx.doi.org/10.1109/ICC40277.2020.9149161.
- [119] J.K. Ray, P. Biswas, R. Bera, S. Sil, Q.M. Alfred, TSN enabled 5G non public network for smart systems, in: 2020 5th International Conference on Computing, Communication and Security, ICCCS, IEEE, Patna, India, 2020, pp. 1–6, http://dx.doi.org/10.1109/ICCCS49678.2020.9277016.
- [120] F. Hamidi-Sepehr, M. Sajadieh, S. Panteleev, T. Islam, I. Karls, D. Chatterjee, J. Ansari, 5G URLLC: Evolution of high-performance wireless networking for industrial automation, IEEE Commun. Stand. Mag. 5 (2) (2021) 132–140, http://dx.doi.org/10.1109/MCOMSTD.001.2000035.
- [121] N. Shibata, P. Zhu, K. Nishimura, Y. Yoshida, K. Hayashi, M. Hirota, R. Harada, K. Honda, S. Kaneko, J. Terada, K.-i. Kitayama, First demonstration of autonomous TSN-based beyond-best-effort networking for 5G NR fronthauls and 1,000+ massive IoT traffic, in: 2020 European Conference on Optical Communications, ECOC, IEEE, Brussels, Belgium, 2020, pp. 1–4.
- [122] A. Mahmood, M.I. Ashraf, M. Gidlund, J. Torsner, J. Sachs, Time synchronization in 5G wireless edge: Requirements and solutions for critical-MTC, IEEE Commun. Mag. 57 (12) (2019) 45–51, http://dx.doi.org/10.1109/MCOM.001. 1900379.
- [123] M. Schüngel, S. Dietrich, L. Leurs, D. Ginthör, S.-P. Chen, M. Kuhn, Advanced grandmaster selection method for converged wired and wireless networks, in: 2021 22nd IEEE International Conference on Industrial Technology, ICIT, vol. 1, IEEE, Valencia, Spain, 2021, pp. 1007–1014.
- [124] O. Seijo, I. Val, M. Luvisotto, Z. Pang, Clock synchronization for wireless timesensitive networking: A march from microsecond to nanosecond, IEEE Ind. Electron. Mag. (2021) 2–10, http://dx.doi.org/10.1109/MIE.2021.3078071.
- [125] H. Shi, A. Aijaz, N. Jiang, Evaluating the performance of over-the-air time synchronization for 5G and TSN integration, in: 2021 IEEE International Black Sea Conference on Communications and Networking (BlackSea-Com), IEEE, Bucharest, Romania, 2021, pp. 1–6, http://dx.doi.org/10.1109/ BlackSeaCom52164.2021.9527833.
- [126] M. Schüngel, S. Dietrich, D. Ginthör, S.-P. Chen, M. Kuhn, Heterogeneous synchronization in converged wired and wireless time-sensitive networks, in: 2021 17th IEEE International Conference on Factory Communication Systems, WFCS, IEEE, Linz, Austria, 2021, pp. 67–74.
- [127] T. Striffler, H.D. Schotten, The 5G transparent clock: Synchronization errors in integrated 5G-TSN industrial networks, in: 2021 IEEE 19th International Conference on Industrial Informatics, INDIN, IEEE, Palma de Mallorca, Spain, 2021, pp. 1–6, http://dx.doi.org/10.1109/INDIN45523.2021.9557468.
- [128] J. Song, M. Kubomi, J. Zhao, D. Takita, Time synchronization performance analysis considering the frequency offset inside 5G-TSN network, in: 2021 17th International Symposium on Wireless Communication Systems, ISWCS, IEEE, Berlin, Germany, 2021, pp. 1–6, http://dx.doi.org/10.1109/ISWCS49558.2021. 9562179.
- [129] 5G Smart Project, 5G E2E Technology to Support Verticals URLLC Requirements, white paper, 2019, URL: https://5gsmart.eu/white-papers/.
- [130] Comcores, Delivering Timing Accuracy in 5G Networks, white paper, 2020, URL: https://www.comcores.com/delivering-timing-accuracy-in-5g-networksieee-1588-ptp-whitepaper/, Accessed: 2021-11-20.
- [131] C. Boles, Shaping the future of the industrial IoT with TSN-over-5G, 2020, URL: https://blogs.intel.com/iot/2020/07/22/shaping-the-future-of-theindustrial-iot-with-tsn-over-5g/#gs.gozm6f, Accessed: 2021-11-20.
- [132] H. Li, D. Li, X. Zhang, G. Shou, Y. Hu, Y. Liu, A security management architecture for time synchronization towards high precision networks, IEEE Access 9 (2021) 117542–117553, http://dx.doi.org/10.1109/ACCESS.2021.3107203.
- [133] X. Wei, X. Xiong, Z. Luo, J. Wang, K. Cheng, An enhanced IEEE1588 clock synchronization for link delays based on a system-on-chip platform, Int. J. Electron. Telecommun. (2021) 289–294, http://dx.doi.org/10.24425/ijet.2021. 135978.
- [134] J. Ansari, T.-s. Hsiao, M.H. Jafari, B. Varga, J. Farkas, I. Moldován, A. Göppert, R.H. Schmitt, 5G enabled flexible lineless assembly systems with edge cloud controlled mobile robots, in: 2022 IEEE 33rd Annual International Symposium on Personal, Indoor and Mobile Radio Communications, PIMRC, 2022, pp. 1419–1424, http://dx.doi.org/10.1109/PIMRC54779.2022.9977496.
- [135] R. Sethi, A. Kadam, K. Prabhu, N. Kota, Security considerations to enable timesensitive networking over 5G, IEEE Open J. Veh. Technol. 3 (2022) 399–407, http://dx.doi.org/10.1109/OJVT.2022.3205014.
- [136] A. Garbugli, L. Rosa, L. Foschini, A. Corradi, P. Bellavista, A framework for TSNenabled virtual environments for ultra-low latency 5G scenarios, in: ICC 2022
 IEEE International Conference on Communications, 2022, pp. 5023–5028, http://dx.doi.org/10.1109/ICC45855.2022.9839193.

- [137] M.-T. Thi, S. Guédon, S.B.H. Said, M. Boc, D. Miras, J.-B. Dore, M. Laugeois, X. Popon, B. Miscopein, IEEE 802.1 TSN time synchronization over Wi-Fi and 5G mobile networks, in: 2022 IEEE 96th Vehicular Technology Conference (VTC2022-Fall), 2022, pp. 1–7, http://dx.doi.org/10.1109/VTC2022-Fall57202. 2022.10012852.
- [138] Y. Li, D. Wang, T. Sun, X. Duan, L. Lu, Solutions for variant manufacturing factory scenarios based on 5G edge features, in: 2020 IEEE International Conference on Edge Computing, EDGE, IEEE, Beijing, China, 2020, pp. 54–58, http://dx.doi.org/10.1109/EDGE50951.2020.00016.
- [139] L. Chettr, R. Bera, Industry 4.0: Communication technologies, challenges and research perspective towards 5G systems, in: Advances in Communication, Devices and Networking, Vol. 662, Springer, Singapore, Singapore, 2020, pp. 67–77, http://dx.doi.org/10.1007/978-981-15-4932-8_9.
- [140] J. Gebert, A. Wich, Alternating transmission of packets in dual connectivity for periodic deterministic communication utilising survival time, in: 2020 European Conference on Networks and Communications (EuCNC), IEEE, Dubrovnik, Croatia, 2020, pp. 160–164, http://dx.doi.org/10.1109/EuCNC48522.2020. 9200911.
- [141] J. Cavazos, 5G is enabling private networks for industry 4.0, 2022, URL: https://www.equipment-news.com/5g-is-enabling-private-networks-forindustry-4-0/, Accessed: 2022-07-13.
- [142] P. Ding, D. Liu, Y. Shen, H. Duan, Q. Zheng, Edge-to-cloud intelligent vehicleinfrastructure based on 5g time-sensitive network integration, in: 2022 IEEE International Symposium on Broadband Multimedia Systems and Broadcasting, BMSB, 2022, pp. 1–5, http://dx.doi.org/10.1109/BMSB55706.2022.9828687.
- [143] J. Huang, L. Feng, F. Zhou, H. Liu, P. Yu, K. Xie, 5G URLLC local deployment architecture for industrial TSN services, in: 2022 International Wireless Communications and Mobile Computing, IWCMC, 2022, pp. 7–11, http://dx. doi.org/10.1109/IWCMC55113.2022.9825113.
- [144] J. Cohen, A coefficient of agreement for nominal scales, Educ. Psychol. Meas. 20 (1) (1960).
- [145] J.L. Fleiss, Measuring nominal scale agreement among many raters, Psychol. Bull. 76 (5) (1971) 378–382.
- [146] M.L. McHugh, Interrater reliability: The kappa statistic, Biochem. Med. 22 (3) (2012) 276–282.
- [147] C. Wohlin, P. Runeson, M. Höst, M. Ohlsson, B. Regnell, A. Wesslén, Experimentation in Software Engineering, Springer, Germany, 2012, http://dx.doi. org/10.1007/978-3-642-29044-2.
- [148] A. Ghosh, A. Maeder, M. Baker, D. Chandramouli, 5G evolution: A view on 5G cellular technology beyond 3GPP release 15, IEEE Access 7 (2019) 127639–127651, http://dx.doi.org/10.1109/ACCESS.2019.2939938.
- [149] A. Jerichow, B. Covell, D. Chandramouli, A. Rezaki, A. Lansisalmi, J. Merkel, 3GPP non-public network security, J. ICT Stand. 8 (1) (2020).
- [150] C.I. Nwakanma, A.P. Anantha, F.B. Islam, J.-M. Lee, D.-S. Kim, 3GPP release-16 for industrial internet of things and mission critical communications, in: 2020 International Conference on Information and Communication Technology Convergence, ICTC, IEEE, Jeju, Korea (South), 2020, pp. 403–406, http://dx. doi.org/10.1109/ICTC49870.2020.9289520.
- [151] S. Jun, Y. Kang, J. Kim, C. Kim, Ultra-low-latency services in 5G systems: A perspective from 3GPP standards, ETRI J. 42 (2020) 721–733.
- [152] M. Wollschlaeger, T. Sauter, J. Jasperneite, The future of industrial communication: Automation networks in the era of the internet of things and industry 4.0, IEEE Ind. Electron. Mag. 11 (1) (2017) 17–27, http://dx.doi.org/10.1109/ MIE.2017.2649104.
- [153] S.S. Craciunas, R.S. Oliver, An overview of scheduling mechanisms for timesensitive networks, in: Proceedings of the Real-Time Summer School LÉcole dÉté Temps Réel, 2017.
- [154] S. Samii, H. Zinner, Level 5 by layer 2: Time-sensitive networking for autonomous vehicles, IEEE Commun. Stand. Mag. 2 (2) (2018) 62–68, http: //dx.doi.org/10.1109/MCOMSTD.2018.1700079.
- [155] L. Deng, G. Xie, H. Liu, Y. Han, R. Li, K. Li, A survey of real-time ethernet modeling and design methodologies: From AVB to TSN, ACM Comput. Surv. 55 (2) (2022) http://dx.doi.org/10.1145/3487330.
- [156] B. Briscoe, A. Brunstrom, A. Petlund, D. Hayes, D. Ros, I.-J. Tsang, S. Gjessing, G. Fairhurst, C. Griwodz, M. Welzl, Reducing internet latency: A survey of techniques and their merits, IEEE Commun. Surv. Tutor. 18 (3) (2016) 2149–2196, http://dx.doi.org/10.1109/COMST.2014.2375213.
- [157] J.W. Won, J.M. Ahn, 3GPP URLLC patent analysis, ICT Express (2020) http: //dx.doi.org/10.1016/j.icte.2020.09.001.
- [158] B. Alzahrani, N. Fotiou, Enhancing internet of things security using softwaredefined networking, J. Syst. Archit. 110 (2020) 101779, http://dx.doi.org/10. 1016/j.sysarc.2020.101779.
- [159] M.M. Islam, M.T.R. Khan, M.M. Saad, D. Kim, Software-defined vehicular network (SDVN): A survey on architecture and routing, J. Syst. Archit. 114 (2021) 101961, http://dx.doi.org/10.1016/j.sysarc.2020.101961.

- [160] Z. Zaidi, V. Friderikos, Z. Yousaf, S. Fletcher, M. Dohler, H. Aghvami, Will SDN be part of 5g? IEEE Commun. Surv. Tutor. 20 (4) (2018) 3220–3258.
- [161] B. Ai, A.F. Molisch, M. Rupp, Z.-D. Zhong, 5G key technologies for smart railways, Proc. IEEE 108 (6) (2020) 856–893, http://dx.doi.org/10.1109/ JPROC.2020.2988595.
- [162] M.A. Ojewale, P.M. Yomsi, B. Nikolić, Worst-case traversal time analysis of TSN with multi-level preemption, J. Syst. Archit. 116 (2021) 102079, http://dx. doi.org/10.1016/j.sysarc.2021.102079, URL: https://www.sciencedirect.com/ science/article/pii/S1383762121000667.
- [163] I. Parvez, A. Rahmati, I. Guvenc, A.I. Sarwat, H. Dai, A survey on low latency towards 5G: RAN, core network and caching solutions, IEEE Commun. Surv. Tutor. 20 (4) (2018) 3098–3130, http://dx.doi.org/10.1109/COMST. 2018.2841349.
- [164] M.A. Khan, H.E. Sayed, S. Malik, T. Zia, J. Khan, N. Alkaabi, H. Ignatious, Level-5 autonomous driving—Are we there yet? A review of research literature, ACM Comput. Surv. 55 (2) (2022) http://dx.doi.org/10.1145/3485767.
- [165] J. Navarro-Ortiz, P. Romero-Diaz, S. Sendra, P. Ameigeiras, J.J. Ramos-Munoz, J.M. Lopez-Soler, A survey on 5G usage scenarios and traffic models, IEEE Commun. Surv. Tutor. 22 (2) (2020) 905–929, http://dx.doi.org/10.1109/ COMST.2020.2971781.
- [166] H. Jiang, G. Gui, Channel Modeling in 5G Wireless Communication Systems, first ed., Springer Publishing Company, Incorporated, Springer, 2019.
- [167] A. Recse, R. Szabo, B. Nemeth, Elastic resource management and network slicing for IoT over edge clouds, in: ACM Proceedings of the 10th International Conference on the Internet of Things, IoT '20, Association for Computing Machinery, New York, NY, USA, 2020, http://dx.doi.org/10.1145/3410992. 3411015.
- [168] A. Kaloxylos, Network slicing for 5G networks: A survey and an analysis of future trends, in: Proceedings of the 21st Pan-Hellenic Conference on Informatics, in: PCI 2017, Association for Computing Machinery, New York, NY, USA, 2017, http://dx.doi.org/10.1145/3139367.3139392.
- [169] G. Blanc, N. Kheir, D. Ayed, V. Lefebvre, E.M. de Oca, P. Bisson, Towards a 5G security architecture: Articulating software-defined security and security as a service, in: ACM Proceedings of the 13th International Conference on Availability, Reliability and Security, in: ARES 2018, Association for Computing Machinery, New York, NY, USA, 2018, http://dx.doi.org/10.1145/3230833. 3233251.
- [170] S. Scanzio, L. Wisniewski, P. Gaj, Heterogeneous and dependable networks in industry – A survey, Comput. Ind. 125 (2021) http://dx.doi.org/10.1016/ j.compind.2020.103388.



Zenepe Satka is a Ph.D. candidate at Mälardalen University in Sweden. She is part of the Heterogeneous System -Hardware Software Co-design (HERO) research group at Mälardalen University. Her main research interests include real-time communication, real-time distributed systems, and timing analysis of distributed embedded systems. She holds a B.Sc. degree and her second M.Sc. degree in Electronic Engineering from the Polytechnic University of Tirana in Albania. She received an M.Sc. degree in Computer Science with a specialization in Embedded Systems from Mälardalen University, Sweden, in 2020, where she is pursuing a Ph.D. in Computer Science.



Mohammad Ashjaei is an Associate Professor in the Complex Real-Time Systems (CORE) and the Heterogeneous Systems - Hardware Software Co-design (HERO) research groups at Mälardalen University in Sweden. Mohammad has received his Ph.D. degree in Computer Science in November 2016 from Mälardalen University. His main research interests include real-time systems, real-time distributed systems, scheduling algorithms on networks and processors, schedulability analysis techniques, resource reservation and reconfiguration mechanisms for real-time networks. He is also giving lectures on various topics related to embedded systems and data communication networks. He is a PC member and referee for several international conferences and journals, including IEEE Transactions on Industrial Informatics (TII), IEEE Transactions on Industrial Electronics (TIE), Elsevier's Journal of Systems Architecture, IEEE Transactions on Network and Services, Journal of Cloud Computing, and ACM Computing Surveys. He has organized and chaired several special sessions and workshops at international conferences such as ETFA.



Hossein Fotouhi is an assistant professor in the School of Innovation, Design, and Engineering at Mälardalen University currently. He received his degree in Electrical Electronics Engineering in 2004. He obtained his Master of Science in Communication Network Engineering in 2009 from University Putra Malaysia. Hossein did his Ph.D. research in CISTER Research Unit from July 2009 till April 2015. His research interests are the Internet of Things, sensor networks, wireless communication, and Fog computing. He has been the guest editor at MDPI Sensors, MDPI JSAN, and Hindawi journals. He has publications in several top venues, including EWSN Conference, IEEE TMC, Elsevier AdHoc Networks Elsevier COMCOM IEEE ACCESS MDPI Sensors, and MDPI JSAN. He has received a very competitive national grant, called the VR starting grant, from the Swedish Research Council. For more information see http://www.es.mdh.se/staff/2992-Hossein Fotouhi.



Masoud Daneshtalab is currently a Prof. at Mälardalen University (MDU), and co-leading the Heterogeneous System research group. He is on the Euromicro board of directors, an editor of the MICPRO journal, and has published over 200 refereed papers. His research interests include HW/SW co-design, reliability, and deep learning acceleration.



Mikael Sjödin is since 2006 a professor of real-time systems at Mälardalen University, Sweden. He is leading the research group Model-Based Engineering of Embedded Systems which focuses on the development of methods and tools for model-based engineering of embedded systems. Including: models for architectural and behavioral descriptions of systems and requirements for systems, techniques for analyzing and transforming models, and runtime architectures for resource-efficient, predictable embedded systems. Since 2012 he is the research director for Embedded Systems, a research environment with 200 active researchers at Mälardalen University. Between 2000 and 2006 Mikael held numerous industrial positions and worked as a software architect and project manager. Mikael received his Ph.D. and M.Sc. in computer science from Uppsala University in 2000 and 1995 respectively.



Saad Mubeen is a Professor at Mälardalen University, Sweden. He has previously worked in the vehicle industry as a Senior Software Engineer at Arcticus Systems and as a Consultant for Volvo Construction Equipment, Sweden. He received his Ph.D. in Computer Science and Engineering from Mälardalen University Sweden in June 2014. He is a Senior Member of IEEE and a Co-chair of the Subcommittee on In-vehicle Embedded Systems within the IEEE IES Technical Committee on Factory Automation. His research focus is on the model- and component-based development of predictable embedded software, modeling, and timing analysis of in-vehicle communication, and end-to-end timing analysis of distributed embedded systems. Within this context, he has co-authored over 150 publications in peer-reviewed international journals, conferences, and workshops. He has received several awards, including the IEEE Software Best Paper Award in 2017. He is a PC member and referee for several international conferences and journals respectively. He is a guest editor of Elsevier's Journal of Systems Architecture and Microprocessors and Microsystems, IEEE Transactions on Industrial Informatics (TII), ACM SIGBED Review, and Springer's Computing journal. He has organized and chaired several special sessions, tracks, and workshops at international conferences such as IEEE's IECON, ICIT, ETFA, ISORC, DIVERSE, and CRTS, to mention a few. For more information see http://www.es.mdh.se/staff/280-Saad Mubeen